

Adjustments to climate perturbations mechanisms, implications, observational constraints

Article

Published Version

Creative Commons: Attribution 4.0 (CC-BY)

Open Access

Quaas, J., Andrews, T., Bellouin, N. ORCID: https://orcid.org/0000-0003-2109-9559, Block, K., Boucher, O., Ceppi, P., Dagan, G., Doktorowski, S., Eichholz, H. M., Forster, P., Goren, T., Gryspeerdt, E., Hodnebrog, Ø., Jia, H., Kramer, R., Lange, C., Maycock, A. C., Muelmenstaedt, J., Myhre, G., O'Connor, F. M., Pincus, R., Samset, B. H., Senf, F., Shine, K. ORCID: https://orcid.org/0000-0003-2672-9978, Smith, C., Stjern, C. W., Takemura, T., Toll, V. and Wall, C. J. (2024) Adjustments to climate perturbations - mechanisms, implications, observational constraints. AGU Advances, 5 (5). e2023AV001144. ISSN 2576-604X doi: 10.1029/2023AV001144 Available at https://centaur.reading.ac.uk/118501/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1029/2023AV001144

Publisher: Wiley



All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other copyright holders. Terms and conditions for use of this material are defined in the <u>End User Agreement</u>.

www.reading.ac.uk/centaur

CentAUR

Central Archive at the University of Reading

Reading's research outputs online

AGU Advances

REVIEW ARTICLE

10.1029/2023AV001144

Peer Review The peer review history for this article is available as a PDF in the Supporting Information.

Key Points:

- Adjustments impact the Earth energy budget, but also circulation, precipitation and atmospheric structure
- Adjustments are classified into six different mechanisms and act at time scales ranging from seconds to multiple years
- Observational constraints can inform on some aspects of adjustments

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

J. Quaas, johannes.quaas@uni-leipzig.de

Citation:

Quaas, J., Andrews, T., Bellouin, N., Block, K., Boucher, O., Ceppi, P., et al. (2024). Adjustments to climate perturbations—mechanisms, implications, observational constraints. *AGU Advances*, 5, e2023AV001144. https://doi.org/10. 1029/2023AV001144

Received 19 DEC 2023 Accepted 12 JUL 2024

Author Contributions:

Conceptualization: Johannes Ouaas. Timothy Andrews, Karoline Block, Olivier Boucher, Paulo Ceppi, Guy Dagan, Hannah Marie Eichholz, Piers Forster, Tom Goren, Edward Gryspeerdt, Øivind Hodnebrog, Hailing Jia, Ryan Kramer, Charlotte Lange, Amanda C. Maycock, Johannes Mülmenstädt, Gunnar Myhre, Robert Pincus, Bjørn Hallvard Samset, Fabian Senf, Keith P. Shine, Chris Smith, Camilla Weum Stjern, Toshihiko Takemura, Velle Toll, Casey J. Wall Formal analysis: Øivind Hodnebrog, Gunnar Myhre, Biørn Hallvard Samset, Chris Smith, Camilla Weum Stjern

© 2024. The Author(s).

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Adjustments to Climate Perturbations—Mechanisms, Implications, Observational Constraints

Johannes Quaas¹, Timothy Andrews², Nicolas Bellouin^{3,4}, Karoline Block¹, Olivier Boucher⁴, Paulo Ceppi⁵, Guy Dagan⁶, Sabine Doktorowski¹, Hannah Marie Eichholz¹, Piers Forster⁷, Tom Goren^{1,8}, Edward Gryspeerdt⁵, Øivind Hodnebrog⁹, Hailing Jia^{1,10}, Ryan Kramer^{11,12}, Charlotte Lange¹, Amanda C. Maycock⁷, Johannes Mülmenstädt¹³, Gunnar Myhre⁹, Fiona M. O'Connor^{2,14}, Robert Pincus¹⁵, Bjørn Hallvard Samset⁹, Fabian Senf¹⁶, Keith P. Shine³, Chris Smith^{7,17}, Camilla Weum Stjern⁹, Toshihiko Takemura¹⁸, Velle Toll¹⁹, and Casey J. Wall²⁰

¹Leipzig Institute for Meteorology, Leipzig University, Leipzig, Germany, ²Hadley Centre, Met Office, Exeter, UK, ³Department of Meteorology, University of Reading, Reading, UK, ⁴Institut Pierre-Simon Laplace, Sorbonne Université/ CNRS, Paris, France, ⁵Imperial College London, London, UK, ⁶Hebrew University Jerusalem, Jerusalem, Israel, ⁷University of Leeds, Leeds, UK, ⁸Bar Ilan University, Ramat Gan, Israel, ⁹CICERO, Oslo, Norway, ¹⁰SRON, Leiden, The Netherlands, ¹¹Earth Sciences Division, NASA Goddard Space Flight Center, Greenbelt, MD, USA, ¹²GESTAR-II, University of Maryland Baltimore County, Baltimore, MD, USA, ¹³Pacific Northwest National Laboratory, Richland, WA, USA, ¹⁴Department of Mathematics and Statistics, Global Systems Institute, University of Exeter, Exeter, UK, ¹⁵Lamont-Doherty Earth Observatory, Columbia University, New York, NY, USA, ¹⁶Leibniz Institute for Tropospheric Research, Leipzig, Germany, ¹⁹International Institute for Applied Systems Analysis, Laxenburg, Austria, ¹⁸Kyushu University, Fukuoka, Japan, ¹⁹University of Tartu, Tartu, Estonia, ²⁰University of Oslo, Oslo, Norway

Abstract Since the 5th Assessment Report of the Intergovernmental Panel on Climate Change (AR5) an extended concept of the energetic analysis of climate change including forcings, feedbacks and adjustment processes has become widely adopted. Adjustments are defined as processes that occur in response to the introduction of a climate forcing agent, but that are independent of global-mean surface temperature changes. Most considered are the adjustments that impact the Earth energy budget and strengthen or weaken the instantaneous radiative forcing due to the forcing agent. Some adjustment mechanisms also impact other aspects of climate not related to the Earth radiation budget. Since AR5 and a following description by Sherwood et al. (2015, https://doi.org/10.1175/bams-d-13-00167.1), much research on adjustments has been performed and is reviewed here. We classify the adjustment mechanisms into six main categories, and discuss methods of quantifying these adjustments that act beyond the energetic framework, and we propose new ideas to observe adjustments or to make use of observations to constrain their representation in models. Altogether, the problem of adjustments is now on a robust scientific footing, and better quantification and observational constraint is possible. This allows for improvements in understanding and quantifying climate change.

Plain Language Summary Climate change is driven by perturbations to the atmospheric composition, to land use, or by changes of incoming solar radiation. It can be understood energetically by quantifying the perturbation to the Earth energy budget—the instantaneous radiative forcing—and the response of the climate system to this perturbation. This response can be split into feedbacks—mechanisms that act in response to global-mean surface temperature changes—and other processes that act independently of the global-mean surface temperature change. These latter processes are called adjustments. There is also a category of climate-relevant adjustments that is not directly related to the energy budget. This review documents the improved classification, understanding, constraint, and quantification of adjustments. A clearer picture of adjustments allows to better understand and quantify climate change.

1. Introduction

Climate change is caused by perturbations to the state or composition of the Earth system that change the balance between the net solar radiation absorbed and the terrestrial radiation emitted to space by the Earth's atmosphere and surface. Perturbations affecting the radiative budget include changes to atmospheric composition (especially greenhouse gases and aerosols or aerosol precursors), changes in land use, and changes to incoming solar





Methodology: Johannes Quaas, Timothy Andrews, Olivier Boucher, Paulo Ceppi, Guy Dagan, Hannah Marie Eichholz, Piers Forster, Tom Goren, Edward Gryspeerdt, Øivind Hodnebrog, Hailing Jia, Ryan Kramer, Charlotte Lange, Amanda C. Maycock, Johannes Mülmenstädt, Gunnar Myhre, Robert Pincus, Bjørn Hallvard Samset, Fabian Senf, Keith P. Shine, Chris Smith, Camilla Weum Stjern, Toshihiko Takemura, Velle Toll, Casey

J. Wall Visualization: Sabine Doktorowski. Øivind Hodnebrog, Gunnar Myhre, Bjørn Hallvard Samset, Camilla Weum Stjern Writing - original draft: Johannes Quaas, Timothy Andrews, Paulo Ceppi, Guy Dagan, Piers Forster, Edward Gryspeerdt, Øivind Hodnebrog, Ryan Kramer, Gunnar Myhre, Fiona M. O'Connor, Bjørn Hallvard Samset, Fabian Senf, Keith P. Shine, Camilla Weum Stiern, Velle Toll, Casey J. Wall Writing - review & editing: Nicolas Bellouin, Karoline Block, Olivier Boucher, Hannah Marie Eichholz, Tom Goren, Hailing Jia, Charlotte Lange, Amanda C. Maycock, Johannes Mülmenstädt, Robert Pincus, Chris Smith, Toshihiko Takemura, Velle Toll

radiation. The resulting change in the net imbalance of radiative flux is known as the *radiative forcing* \mathcal{F} . Radiative forcing was first identified as a means of quantifying the climate response to perturbations by Ramanathan (1975). Understanding was consolidated through the late 1970s and early 1980s (e.g., Dickinson & Cicerone, 1986; Ramanathan & Dickinson, 1979; Ramanathan et al., 1979) so that the concept of radiative forcing was in widespread use by the time of the first assessment of the Intergovernmental Panel on Climate Change (IPCC, 1990). Ramaswamy et al. (2019) provide further historical context.

A radiative imbalance will induce a surface temperature change as the system warms or cools toward a new equilibrium. The energy imbalance *N* caused by a forcing will be damped by the net climate feedback λ acting on the surface temperature perturbation ΔT_s , which can be described in a linearized form for small perturbations around an equilibrium state in the global mean:

$$\mathbf{V} = \mathcal{F} + \lambda \Delta T_{\rm s}.\tag{1}$$

Equation 1, which applies to the time-integrated global mean, embodies the "forcing-feedback" framework for temperature change (Gregory et al., 2004). The top-of-atmosphere imbalance N is taken up in Earth's climate system, with ocean heat uptake accounting for about 90% of the total (Forster et al., 2021) at present. Implicit in Equation 1 is the assumption that surface temperature change is uniquely determined by the feedback parameter and the forcing regardless of the mechanism inducing the forcing.

A literal interpretation of Equation 1 suggests that \mathcal{F} may be determined as the difference in top-of-atmosphere flux between two radiative transfer calculations that are identical except with respect to the forcing agent being present or not—the "instantaneous radiative forcing" at the top of the atmosphere. \mathcal{F} so computed is, however, an imperfect predictor of eventual temperature change (e.g., Hansen et al., 1997). Especially vexing is that the feedback parameter computed with this definition often depends on the forcing agent.

Computing radiative forcing as the flux change at the tropopause, rather then the top of the atmosphere, increases the degree to which the same forcing magnitude elicits the same feedback and surface temperature response, irrespective of the mechanism causing the forcing (Hansen et al., 1997). Early work showed further that "an improved measure of radiative forcing is obtained by allowing the stratospheric temperature to adjust to a radiative equilibrium profile" (Hansen et al., 1997). Stratospheric temperature adjustment is especially important for forcing by carbon dioxide, which acts to cool the stratosphere (Manabe & Wetherald, 1967), further reducing the terrestrial radiation lost to space (i.e., the stratospheric cooling increases the radiation imbalance *N*). This temperature *adjustment*, which returns the stratosphere to a state of (global-mean) radiative near equilibrium, acts on a more rapid timescale (weeks–months) than the surface temperature response (years to decades), and is independent of surface temperature change. This observation led early investigators (e.g., Ramanathan & Dick-inson, 1979; Ramanathan et al., 1979) to view changes due to the stratospheric temperature adjustment as part of the forcing. Forcing including this adjustment is often reported at the tropopause but, since stratospheric radiative equilibrium requires no radiative flux divergence across the stratosphere, forcing at the tropopause and top-of-atmosphere are identical once the stratospheric temperature adjusts.

Further experience with climate models (starting with Hansen et al. (1997)) has demonstrated that stratospheric temperature adjustments in response to carbon dioxide abd other forcings is one of many *adjustments* to forcing in the climate system. Adjustments are defined as changes in the state or composition of the Earth system, caused by an initial forcing agent, that modify the top-of-atmosphere energy imbalance in the absence of a change in global-mean surface temperature (Andrews & Forster, 2008; Bony & Stevens, 2020; Colman & McAvaney, 2011; Gregory & Webb, 2008; Sherwood et al., 2015). The term is used to note both the change in state and the resulting change in top-of-atmosphere flux. Cloud adjustments occurring in response to arbitrary forcings were recognized more than 15 years ago (Andrews & Forster, 2008; Gregory & Webb, 2008). Cloud adjustments to aerosol forcing as a consequence of aerosol-cloud interactions that was identified even much earlier (Lohmann et al., 2010; Rotstayn & Penner, 2001), are thought to be an important component of total anthropogenic aerosol forcing (Bellouin et al., 2020).

The recognition of the substantial role of adjustments motivated the definition of *effective radiative forcing* (ERF or \mathcal{E} in the notation of Bellouin et al. (2020)) as the sum of the (purely radiative) instantaneous response and any adjustments. Climate model simulations follow Equation 1 more consistently when \mathcal{E} replaces \mathcal{F} , that is, effective

radiative forcing is a better predictor of global temperature change than the instantaneous radiative forcing computed with fixed state and composition. Adjustments are also useful in understanding precipitation responses to forcing (e.g., Richardson et al., 2018).

Here we assess and synthesize the current state of understanding on adjustments, updating the knowledge since the review by Sherwood et al. (2015). Section 2 highlights the evolving conceptual understanding of adjustments including limitations in existing diagnostic techniques, with a more explicit categorization than in Sherwood et al. (2015). The wide range of mechanisms by which adjustments can arise is surveyed in Section 3, highlighting several that have recently come to light. Section 4 takes a closer look at non-radiative adjustments such as atmospheric circulation and precipitation changes, and Section 5 highlights opportunities for constraining adjustments with observations, with a particular focus on aerosol-cloud interactions. Opportunities for developing understanding are highlighted in the final section.

2. Definitions and Framework

Measures of radiative forcing quantify the difference in energy balance between two states. As only one of these states is directly observable, estimates of radiative forcing necessarily rely on modeling. The character of the modeling depends on the choice of forcing metric and introduces uncertainties and constraints.

Instantaneous radiative forcing, as noted above, can be determined directly from radiative transfer calculations for which the physical basis is extremely strong. Stratospheric temperature adjustments may be estimated by assuming that the stratosphere is close to radiative equilibrium and that the resulting radiative heating rates remain constant after forcing. Forcing under this "fixed dynamical heating" assumption can be computed by iteratively adjusting the stratospheric temperature profile in the presence of the forcing agent until it reproduces the radiative heating rates in the unperturbed state (e.g., Fels et al., 1980; Forster et al., 1997). Stratospheric temperature adjustments substantially enhance forcing by carbon dioxide and perturb ozone forcing but have little impact on optically thinner gases such as methane and halocarbons (e.g., Pincus et al., 2020).

In general, however, assessing adjustments requires calculations with more complete models of the climate system. Two general approaches are available for estimating adjustments and the resulting ERF from climate models. One, introduced by Gregory et al. (2004), exploits the linearity of Equation 1 to determine λ as the slope and \mathcal{E} as the intercept of a linear regression of N against ΔT_s in simulations with abruptly increased forcing. The second suppresses ΔT_s in Equation 1 by fixing surface temperature at its control value and measuring the top-of-atmosphere imbalance in the presence of forcing agents (Hansen et al., 2005). Fixing land surface temperatures in climate models has historically been a technical challenge, so current practice is to calculate ERF by fixing sea surface temperatures and sea ice concentrations (Forster et al., 2016). Radiative kernels (Soden et al., 2008) can be applied to correct the ERF estimated with fixed sea surface temperatures to account for land surface temperature changes.

These approaches represent two approximations to a conceptual definition of adjustment. Because global-mean temperature increases even in the first year of a simulation with abrupt forcing, regression (the "Gregory" method), by defining ERF as the imbalance for a state with zero global-mean surface temperature change, requires adjustments to occur on rapid (sub-yearly) timescales. This definition inspired the use of the term *rapid adjustments* though this language is potentially misleading (Forster et al., 2021). Surface temperature change suppression (the "Hansen" approach), in contrast, imposes no constraint on the timescale of adjustments, but fixes surface temperature at each point (rather than only in the global average), potentially suppressing circulations and adjustments caused by local temperature anomalies. In practice the two approaches generally yield similar ERF values for individual models relative to the large inter-model spread (Chung & Soden, 2015; Forster et al., 2016).

The implications of such definitions can be understood by expanding Equation 1:

$$\overline{N}(t) = \overline{\mathcal{F}} + \sum_{i} \overline{A}_{i}(t) + \sum_{j} \overline{\left(\lambda_{A,j}(t)\Delta T_{j}'(t)\right)} + \sum_{k} \lambda_{fb,k} \overline{\Delta T_{s}}(t)$$
(2)

Here the imbalance includes the instantaneous radiative forcing \mathcal{F} damped by radiative feedbacks (k indexes feedback processes including the Planck, surface albedo, water vapor, lapse-rate and cloud feedbacks with feedback parameters $\lambda_{\text{fb},k}$) that occur in response to the global-mean surface temperature change $\overline{\Delta T_s}(t)$. These





Figure 1. Mechanisms of adjustments: Thermodynamical adjustments, land surface temperature adjustments, vegetation and chemical adjustments, cloud microphysical adjustments in response to aerosol perturbations, and circulation adjustments in response to temperature anomalies.

feedbacks may occur across a range of time scales (e.g., Hansen et al., 2023). There are three classes of adjustments.

- Adjustments independent of any surface temperature changes. This is the best-defined class of adjustments. In Equation 2, it is the second term (A_i) , where *i* indicates the summation over different adjustment mechanisms.
- Adjustment mechanisms involving local temperature changes ΔT' with (near-)zero global-mean temperature change. These are described by the third term. This term is often somewhat problematic in interpretation and in disentanglement from the adjustments A_i and from the feedbacks. We refer to these later as "land surface temperature adjustments" and "dynamical adjustments."
- Adjustment processes that are relevant for climate but not for the Earth radiation budget. These do not appear in the above equation and thus are not damping or amplifying climate change. They are thus outside the forcing-feedback framework. It seems useful to nevertheless also consider them when discussing mechanisms acting independent of global-mean surface temperature change (Section 4).

The terms for the adjustments in Equation 2 are assumed to add up linearly in analogy to the feedback terms. An interesting manifestation of adjustments is the effect of orbital forcing that leads to ice age cycles. Over time scales of tens to hundreds of millennia, large climate changes such as the glacial—interglacial ages are caused by periodic change in properties related to the Earth's orbit around the Sun such as its eccentricity, obliquity and precession. For the latter two, only the seasonal and latitudinal distribution of incident sunlight is affected but not the global annual mean. This implies that glacial-interglacial temperature changes arise almost entirely from adjustments (the growth or decay of ice sheets) which increase or decrease surface albedo. The differential absorption of sunlight could also drive changes in the Hadley circulation and in monsoons (Erb et al., 2015). While this is a rather extreme case in the context of adjustments and climate change since predindustrial times, it serves to highlight the limitations of the linearized forcing-feedback framework in Equation 1.

We note tangentially that Equation 2 has wider applications. Simple physical emulators of climate change adopt this energy budget framework to model global temperature projections in response to radiative forcing or emissions scenarios, with typically a two or three layer ocean model to represent different layers or timescales (e.g., Geoffroy et al., 2013). Recent characterizations of such models link emissions to ERF, explicitly accounting for adjustments within their ERF evaluation (Smith, Forster, et al., 2018).

3. Adjustment Mechanisms

Different drivers of climate change may cause different adjustment mechanisms (Figure 1). The mechanisms discussed below are included in the second term on the right-hand-side of Equation 2 (thermodynamical,





Figure 2. The relative importance (in %) of adjustment processes to ERF for an increase in 10 different climate drivers. These include atmospheric concentrations of greenhouse gases (CO_2 , methane/ CH_4 , nitrous oxide/ N_2O , halocarbons/CFC-11 and CFC-12, and ozone/ O_3), absorbing (BC) and scattering (SO_4) aerosol as well as volcanic aerosol, and incoming solar radiation changes. Note that SO_4 and volcanic aerosols have negative ERF. Numbers are taken from Hodnebrog et al. (2020), Marshall et al. (2020), Skeie et al. (2020), and Smith, Kramer, et al. (2018).

vegetation, microphysical and chemical adjustments) and in the third term (land surface and dynamical adjustments). Most of the former mechanisms (in particular, microphysical and chemical adjustments, but also vegetation adjustments to a large extent) are due to the nature of the driver of the forcing, rather than due to the instantaneous radiative forcing itself.

- 1. *Thermodynamical adjustments*. Some climate drivers interact with solar or terrestrial radiation to cause heating or cooling within the atmosphere. This can trigger further changes to surface fluxes, clouds, water vapor and precipitation.
- 2. Vegetation adjustments. Vegetation reacts to elevation in CO_2 concentrations (fertilizing effect), to tropospheric ozone (damaging effect), and to aerosols (increase in diffuse radiation) altering surface energy fluxes, emissions of radiatively active atmospheric constituents (e.g., dust) and emissions of ozone and aerosol precursors (e.g., biogenic volatile organic compounds (BVOCs)).
- 3. *Microphysical adjustments*. Clouds and precipitation respond to changes in the abundance of cloud condensation nuclei (CCN) and ice nucleating particles (INPs), altering their microphysical state and thus their radiative properties in response to anthropogenic aerosol and aerosol precursor emissions.
- 4. *Chemical adjustments*. Atmospheric chemistry acting on anthropogenic emissions of chemically active trace gases leads to changes in oxidizing capacity and other forcing constituents (e.g., ozone).
- 5. Land surface adjustments. Heating or cooling of the surface by for example, atmospheric aerosols or greenhouse gases, aerosol-cloud interactions or surface albedo changes, may lead to adjustments by modifying the surface energy budget and surface fluxes even with no change in global-mean surface temperature or, depending on definition, if changes in surface temperature are limited to land only.
- Dynamical adjustments. Anomalies in atmospheric circulations are induced by changes in surface and atmospheric temperature patterns, with consequences for the radiation budget for example, by changes in cloud patterns.

These six mechanisms of adjustments will be discussed in the following subsections.

Another way of analyzing adjustments is by separating their effects on the different state variables relevant for radiation, as is commonly done in feedback analysis (e.g., Bony et al., 2006). Several studies have investigated adjustments to different climate drivers in this way, by using radiative kernels (Hodnebrog et al., 2020; Marshall et al., 2020; Skeie et al., 2020; Smith, Kramer, et al., 2018; Smith et al., 2020). Figure 2 illustrates how they diagnose the adjustment processes, and their relative importance compared to the ERF. The surface temperature and albedo adjustments are weak for all drivers shown in Figure 2. BC stands out with a relatively large change for tropospheric temperature, water vapor and cloud adjustments. Stratospheric temperature adjustment is important for CO_2 , but also for other drivers such as ozone, BC and changes to insolation.



Figure 3. The relative importance (in %) of adjustment processes to ERF for four different mechanisms. The values are taken from the literature. Adjustments to increases in atmospheric CO_2 concentrations, to changes in aerosol concentrations, and the chemical adjustments contributing to the effective forcing by methane and NO_x emissions are considered as examples. Values are from Doutriaux-Boucher et al. (2009, DB09), Zarakas et al. (2020, Z20), Mahowald (2011, M11), Bellouin et al. (2020, B20), Szopa et al. (2021, S21), and Andrews et al. (2021, A21).

Figure 2 addresses the thermodynamic adjustments. For the other adjustment mechanisms, no systematic multimodel assessment is available yet. A compilation of some studies on relative magnitudes of the different adjustment mechanisms for different drivers is shown in Figure 3.

A summary of the level of scientific understanding and the level of quantification is provided in Table 1.

3.1. Thermodynamical Adjustments—Adjustments of Temperature, Humidity, and Cloud Profiles

Thermodynamical adjustments are the changes in atmospheric heating rates and, subsequently, in temperature, humidity and cloudiness, in response to absorption or emission of radiation by climate drivers. Figure 4 illustrates how atmospheric absorption causes adjustments and feedbacks on different timescales following increases in three very distinct climate drivers, namely CO₂, BC and sulfate aerosols (based on results from Stjern, Forster, et al. (2023)). CO_2 causes an immediate radiative cooling in the stratosphere and a weaker radiative heating in the troposphere. The resulting stratospheric temperature reductions occur on a timescale mainly from weeks to months, while most of the tropospheric temperature increase is slower. The CO₂driven temperature adjustments cause changes in atmospheric static stability and in humidity, triggering cloud changes in the lower and upper troposphere within days to weeks. A further effect is the change in cloud-top radiative cooling in response to altered greenhouse gas concentrations with consequences for cloud lifetime (Bretherton, 2015). Larger cloud changes, on much longer timescales, follow as surface temperaturedriven feedbacks increase with time (see Section 3.5). Whereas CO_2 mostly interacts with longwave radiation, BC interacts with solar radiation. The solar absorption by BC causes a rapid temperature increase in the atmosphere, with a vertical structure depending on the vertical distribution of BC (Stjern et al., 2017). Large cloud reductions in most of the troposphere and cloud increases near the surface occur within days to weeks. Sulfate aerosols have initially a small impact on the absorption of radiation within the atmosphere and adjustments are very weak. Changes to the incoming solar radiation similarly mostly act on the surface energy budget but have some effect on atmospheric heating in the troposphere via changes in the ultraviolet (Gray et al., 2009).

R	ı	G	н	т	s	L	I	N	к	0
C * 1		1.8.3		1.0			κ.			

Summary of Adjustment Mechanisms		
Adjustment	Level of scientific understanding	Level of quantification
Thermodynamical	Medium. (T)—Clear concept for radiative heating due to absorption, less clear for cloudiness changes. (M)—Qualitative agreement across models for large-scale averages for radiative heating, broadly also for cloud changes. (O)—Observational evidence clear for stratosphere, but not yet for the troposphere	Medium. Large model diversity for cloud adjustments. However, no observational estimate yet
Vegetation	Low. Basic physiological responses of vegetation to CO ₂ are understood but radiative effects and further climate responses are not well understood. (M) Very limited analysis of the contribution of vegetation changes to radiative forcing exists and uncertainties have not been evaluated. (O) Observations suggest that vegetation influences Earth's radiative budget but understanding is limited and not specific to radiative forcing	Low. Estimates of the contribution of vegetation changes to radiative forcing has only been diagnosed in a few cases and no quantification of uncertainties or inter-model spread have been conducted
Cloud microphysical	Medium. (T)—Multiple mechanisms explained, partly of opposite sign. Less clear but likely much smaller signal for mixed- and ice-phase clouds. (M)—Models tend to agree on increases in horizontal and vertical extent as aerosol concentrations increase. (O)—Mixed results on vertical extent	Low. The magnitude, and in some cases even the sign, of microphysical adjustments, is still very uncertain. Observational estimates are available but no consensus reached yet
Chemical	Medium in general, with uncertainties in secondary aerosol effects. (T)—Good theoretical understanding of chemical responses to emissions. (M)—Good qualitative agreement on ozone and methane lifetime adjustments; Poor agreement for other trace gases (e.g., VOCs) and secondary aerosol effects. (O)—Strong evidence in some cases (e.g., stratospheric ozone); Inconsistencies in tropospheric ozone observed trends	Medium. Good quantitative agreement in some cases (e.g., ozone ERF); Large model diversity in aerosol-mediated cloud adjustment from oxidizing capacity changes. Observational constraint feasible in some cases (e.g., ozone adjustment from halocarbon perturbation), no estimate of aerosol response to oxidizing capacity
Land surface temperatures	Medium. (T) Clear understanding of thermodynamic underpinnings as surface temperature warms with a positive instantaneous forcing but (M) uncertainty in the magnitude of the direct surface temperature adjustments remains and uncertainty of indirect adjustments associated with the temperature change has not been explored. (O) No observational support yet. Isolating these adjustments may not be possible in observations	Low. Only the total surface temperature adjustment is regularly quantified. Isolation of the contributions from the land temperature change to other adjustments has been infrequent. No observational quantification exists
Dynamical	Low in general, though some mechanisms better understood. (T)— Good understanding of circulation responses to land-sea contrast. Many mechanisms for eddy-driven circulation shifts have been proposed, but exact mechanism uncertain. (M)— Generally considerable model uncertainty in circulation adjustments, although most models simulate a poleward shift of the jet and storm track for CO ₂ . (O)—No observational evidence yet	Low. Qualitative model agreement on CO ₂ adjustments, but greater uncertainties for aerosols. No observational estimate yet

7 of 22



AGU Advances





3.2. Vegetation Adjustments

Vegetation may respond to various climate drivers. Due to the impact of vegetation on albedo and on atmospheric humidity, as well as CO₂ concentrations, but also the emission of soil dust and BVOCs, this implies a radiative adjustment. Increases in atmospheric ozone concentrations may damage vegetation (Reich & Amundson, 1985) and thus reduce the vegetation effect on climate. Increases in aerosol lead to enhanced fraction of diffuse versus direct radiation, implying more efficient photosynthesis and thus reduction in atmospheric CO₂ concentrations (Mercado et al., 2009) with further consequences on albedo.

A further important adjustment process is due to the physiological response. As CO₂ concentrations increase, leaf stomatal conductance is reduced. This suppresses the amount of water transpiration that occurs when the plant absorbs the excess CO_2 (e.g., Leakey et al., 2009). In consequence, surface latent and sensible heat fluxes are affected, and thus boundary layer temperature and humidity, and subsequently cloud cover, each impacting the CO₂ ERF as adjustments (Andrews et al., 2012; Doutriaux-Boucher et al., 2009). Likewise, increased CO₂ fertilization by plants generally decreases surface albedo by increasing leaf area, leading to associated radiative changes (Bala et al., 2006; Zarakas et al., 2020). Increased CO_2 fertilization may also decrease dust emissions through changes in bare soil fraction, but there is no consensus across models on the sign of that effect (e.g., Thornhill, Collins, Olivié,

et al., 2021). Vegetation also responds to precipitation changes (Section 4), but such effects have not yet been investigated very much.

Both the reduction in stomatal conductance and the increased CO_2 fertilization are a direct consequence of the CO_2 concentration increase, independent of T_s change. However, physiological changes also induce land temperature change as the suppression of transpiration by stomata closure reduces evaporational cooling at the surface. This implies further adjustments as discussed below (Section 3.5).

Physiological changes to other forcing agents also may be considered (e.g., fertilization by dust, or vegetation changes in response to microphysical precipitation changes) but there is no evidence these are non-negligible in terms of their impact on the Earth energy budget. We note some of these physiological processes, such as changes in leaf area index or plant functional type, evolve over generations of plants and thus on time scales of years to decades.

3.3. Cloud and Precipitation Microphysical Adjustments to Aerosol Perturbations

In addition to their interaction with radiation (Section 3.1), aerosols serve as nuclei for the formation of cloud droplets and ice crystals. Increases in aerosol concentration typically result in increases in the cloud droplet number concentration (N_d ; Squires, 1952), which leads to an increase in cloud reflectivity (Twomey, 1974). Aerosol perturbations may also lead to changes in ice crystal number concentration, but only a small fraction of anthropogenic aerosols act as ice nucleating particles (Burrows et al., 2022; Kanji et al., 2020; Murray et al., 2012; Vergara-Temprado et al., 2018).

The changes to the hydrometeor size distribution in response to aerosol concentration changes and the resultant radiative effect is quasi-instantaneous and thus is the IRF due to aerosol-cloud interactions (IRFaci). It is noted that one could take the view that the IRFaci is an adjustment to the aerosol perturbation, but the community, including IPCC, has chosen to regard it as an IRF.

Changes to the hydrometeor size distributions produce changes in cloud microphysical processes and in turbulence. In liquid clouds, this includes a modification of precipitation formation (Albrecht, 1989), droplet sedimentation and evaporation (Ackerman et al., 2004; Bretherton et al., 2007) and turbulent entrainment (Small et al., 2009). These changes in processes lead to adjustments through modifications in cloud macrophysical properties, such as cloud fraction (Albrecht, 1989) and water path (Pincus & Baker, 1994). These adjustments act on a timescale of hours to days (Dagan et al., 2017; Glassmeier et al., 2021; Gryspeerdt et al., 2021; Seifert et al., 2015). With strong dependence on cloud type, they exhibit large regional variation (Bellouin et al., 2020) and temperature dependence, with the potential to modify cloud feedbacks and climate sensitivity (Dagan, 2022a; Murray-Watson & Gryspeerdt, 2022; Zhang et al., 2022).

Microphysical adjustments are hypothesized to be important also in mixed-phase and ice clouds (DeMott et al., 2010; Fan et al., 2013; Lohmann, 2002, 2017). The adjustments in mixed-phase and ice clouds can be caused by mechanisms different from the adjustment mechanisms in liquid water clouds. Anthropogenic INPs can lead to the glaciation of supercooled cloud droplets as INPs are needed for ice nucleation at temperatures warmer than about -36° C (DeMott et al., 2010). Glaciation can lead to changes in precipitation, cloud lifetime and radiative fluxes (Lohmann, 2002; Storelvmo et al., 2008). Anthropogenic INPs and associated cloud adjustments could also impact cloud phase feedback to global warming as higher INP concentrations would mean more ice clouds that can be transformed into liquid clouds with global warming (Murray et al., 2021). In addition, INPs could alter the properties of cirrus clouds (DeMott et al., 2010; Kärcher & Lohmann, 2003). Finally, additional CCN might considerably affect deep convective clouds, associated latent heating, precipitation and radiative fluxes through various mechanisms (Fan et al., 2013; Koren et al., 2010; Rosenfeld et al., 2008; E. Williams et al., 2002).

While the magnitude of IRFaci remains uncertain (Bellouin et al., 2020), the processes driving it have a moderate level of understanding and confirmation by observations. The adjustments to aerosol-cloud interactions are much less well understood. With a complex array of interacting processes (Stevens & Feingold, 2009), the magnitude (and in some cases the sign) of these adjustments is uncertain (Bellouin et al., 2020) and differs between observations and global models (Bender et al., 2019; Gryspeerdt et al., 2020; Malavelle et al., 2017; Possner et al., 2020; R. Wood, 2007). Adjustments in mixed-phase and ice clouds are even less well understood and the

confidence in the magnitude of these adjustments remains low (Bellouin et al., 2020). Some quantitative assessments of cloud microphysical adjustments are discussed in Section 5.

3.4. Chemical Adjustments

Atmospheric chemistry plays a role when reactive compounds are emitted to the atmosphere. These emitted compounds may exert an IRF (e.g., halocarbons) or may be relatively radiatively inactive (e.g., nitrogen oxides; NO_x). In both cases, chemical adjustments following their emission alter the abundance of radiatively active trace gases (e.g., O_3) and lead to additional perturbations to the Earth's radiation balance.

Anthropogenic nitrous oxide (N₂O) emissions exert an IRF through absorption of terrestrial outgoing radiation (O'Connor et al., 2021), with both chemical adjustments—arising from a reduction in stratospheric O_3 and a shortening of the CH₄ lifetime—and cloud adjustments having consequences for the radiation budget (O'Connor et al., 2021; Szopa et al., 2021; Thornhill, Collins, Kramer, et al., 2021). A similar effect is found with halocarbon emissions (Szopa et al., 2021).

The ERF from anthropogenic CH₄ emissions is also strongly modulated by chemical adjustments (Szopa et al., 2021). CH₄-driven increases to its own lifetime, tropospheric O_3 , and stratospheric water vapor contribute substantially to the CH₄ ERF (Myhre et al., 2013; Shindell et al., 2005). CH₄ lifetime and abundance are also modulated by chemical adjustments from other emitted compounds (Thornhill, Collins, Kramer, et al., 2021). For example, NO_x reduces CH₄ lifetime (resulting in a negative NO_x ERF overall, despite a positive tropospheric O_3 chemical adjustment to NO_x; Szopa et al., 2021, their Figure 6.12), while emissions of carbon monoxide (CO) and volatile organic compounds (VOCs) increase CH₄ lifetime (contributing approximately 30% to their positive ERF; Szopa et al., 2021).

Anthropogenic changes in stratospheric and tropospheric O_3 arise from chemical adjustments. Of the resulting ERF, 45% are attributable to anthropogenic CH₄, a third due to CO and VOCs, and a quarter due to NO_x (Stevenson et al., 2013), but with some offsetting from the impact of halocarbons on stratospheric O_3 and subsequent reduction in tropospheric O_3 (Szopa et al., 2021). Stratospheric ozone also responds to changes in incoming solar radiation, particularly at ultraviolet wavelengths, as a result of changes in photochemistry. This reduces the modeled surface warming impact of solar perturbations by about one third (Chiodo & Polvani, 2016).

Changes in tropospheric O_3 and hydroxyl (OH) radical production arising from anthropogenic emissions can also affect secondary aerosol production and aerosol size distributions. This may reduce the ERF from aerosol-cloud interactions by 20% (Karset et al., 2018). It may also offset the positive tropospheric O_3 chemical adjustment from anthropogenic NO_x (O'Connor et al., 2021) and change the sign of the cloud adjustment attributable to anthropogenic CH_4 from negative to positive (O'Connor et al., 2022).

3.5. Land Surface Temperature Adjustments

The IRF due to most climate drivers is spatially heterogeneous. Thus, even for unchanged global-mean nearsurface temperatures, a pattern of temperature changes occurs. Beyond this, in practice, the community diagnoses ERF using simulations where sea surface temperatures are fixed (fixed-SST), but land temperatures are free to respond, largely due to technical hurdles in fixing surface temperatures over land (Forster et al., 2016). In such cases, the ERF is diagnosed from conditions where $\overline{\Delta T_s} \neq 0$, even if inconsistent with the formal definition. In particular over land, temperatures respond at rapid time scales. The land temperature change (ΔT_L) occurs within several days after the imposed perturbation (quantified at ~5 days by Dong et al. (2009)). Using high-temporalresolution output from models contributing to the Precipitation Driver Response Model Intercomparison Project (Myhre et al., 2017), Stjern, Forster, et al. (2023) find the land begins to cool to an extent distinguishable from the inter-model variability within 1 day after a five-fold increase in SO₄. After a doubling of CO₂, the land begins to warm within just a few hours. These temperature changes imply non-negligible radiative adjustments.

The $\Delta T_{\rm L}$ primarily influences the ERF by directly modifying longwave emissions (Planck effect or surface temperature in Figure 2). This direct effect can be substantial, accounting for nearly all of the tropospheric temperature adjustment for perturbations in CO₂, CH₄, and SO₄ (Smith, Kramer, et al., 2018). The $\Delta T_{\rm L}$ also causes indirect radiative changes as the Earth system rapidly responds to the temperature change. These effects are not separable in coupled or fixed-SST simulations, so they have remained largely undiagnosed. Andrews et al. (2021) provide the most comprehensive attempt to date, isolating all $\Delta T_{\rm L}$ -induced radiative effects using

simulations by Ackerley et al. (2018) with both SSTs and land temperatures fixed in a single general circulation model (following Shine et al. (2003)). In the model evaluated by Andrews et al. (2021), the ΔT_L -induced reduction in the CO₂ ERF manifests itself through increased emission of terrestrial radiation from the land surface and troposphere, reduced land albedo, increased moisture and through cloud adjustments associated with the land warming. When compared against traditional fixed-SST simulations, they find land warming reduces the ERF by 1 W m⁻² for a quadrupling of CO₂, or roughly 14% of the total ERF. This suggests a proportional underestimate of equilibrium climate sensitivity occurs when fixed-SST simulations are used to calculate the ERF, highlighting the importance of understanding and quantifying the radiative effects of ΔT_L . As a solution that allows for the freedom of the model to develop diurnal cycles and to respond to synoptic-scale weather variability, the temperature could be fixed at a soil level below the surface. Alternative approaches use a model's feedback parameter scaled by ΔT_L (Hansen et al., 2005), or use radiative kernels to subtract out portions of the adjustments that are likely to be ΔT_L related (Smith et al., 2020; Tang et al., 2019; Thornhill, Collins, Kramer, et al., 2021). However, simulations with fixed land temperatures provide the most accurate estimate of the ERF and we encourage additional groups to perform these experiments.

Some of the $\Delta T_{\rm L}$, and its associated radiative effects, however, occur due to local processes and are independent of a change in $\overline{T_{\rm s}}$ (i.e., at global-mean $\Delta \overline{T_{\rm s}} = 0$). This component of the land surface temperature adjustment thus is a radiative adjustment in the strict sense. However, it has not yet been quantified in detail. Cloud adjustments arise from reduced stability and cloudiness over land (a positive adjustment) counteracted by an increase in lower tropospheric stability and low-altitude cloudiness over the oceans (negative adjustment). In total these cloud adjustments act to reduce the ERF, but the magnitude and even sign is model dependent. The contrasting changes over land and ocean suggest that the land warming causes land-sea circulation changes (see Section 3.6).

3.6. Dynamical Adjustments

Dynamical adjustments arise from spatial heterogeneity in surface and atmospheric temperature changes (third term on the right-hand side of Equation 2). Non-uniform aerosol forcing can cause changes in regional and global circulation, as several studies have shown (Bollasina et al., 2011; Chemke & Dagan, 2018; Ganguly et al., 2012; Johnson et al., 2019; G. Persad et al., 2023; Roeckner et al., 2006; Xie et al., 2022). For example, land surface cooling induced by aerosols can alter monsoon circulations even in a situation where SSTs are held fixed (Bollasina et al., 2011; Dong et al., 2015; Ganguly et al., 2012; Guo et al., 2013; Voigt et al., 2017). Adjustments to CO_2 also involve a land-sea contrast and a change in the monsoon circulations, which in some regions are opposite in sign to the changes driven by subsequent sea surface warming (Shaw & Voigt, 2015).

In addition, absorbing aerosols lead to atmospheric heating, which can modify large-scale atmospheric circulations (Johnson et al., 2019; Sand et al., 2020; T. Wood et al., 2020; Xie et al., 2020), with the perturbation's location being a crucial factor in this modulation (Dagan et al., 2019; A. I. Williams et al., 2022). Specifically, tropical atmospheric heating can cause direct thermally driven circulations because of the absence of a significant Coriolis force, while the response in the extra-tropics is more locally focused (Dagan et al., 2019). Furthermore, the strength of these large-scale circulation adjustments varies depending on the background circulation at the site of the added aerosols (A. I. Williams et al., 2022). Consequently, the magnitude and even sign of the generated ERF can be influenced by the perturbation's location (G. G. Persad & Caldeira, 2018; A. I. Williams et al., 2022). The potential implications of large-scale circulation adjustments for ACI are much more challenging to elucidate due to the wide range of scales involved and the need to consider microphysical processes while accounting for large-scale circulation. As a result, little is known about ACI-driven large-scale circulation adjustments. Nevertheless, idealized convective-permitting simulations that use parameterized (Abbott & Cronin, 2021; Dagan, 2022b) or directly resolved (Dagan et al., 2023) large-scale vertical velocity have shown that a local ACI perturbation could strengthen large-scale circulations and subsequently drive an increased cloudiness that results in stronger (more negative) ERF (Dagan et al., 2023).

Changes in the eddy-driven circulation, including the width of the Hadley cells, as well as the position and strength of the midlatitude jet and storm tracks and the southern hemisphere polar jet, are also subject to adjustments to forcing (Grise & Polvani, 2014, 2016; Morgenstern et al., 2020; Staten et al., 2014; T. Wood et al., 2020). For example, the warming effects by additional BC lead to a widening of the tropical belt and poleward shift of the midlatitude storm tracks, while additional scattering aerosol has the opposite effect (Johnson et al., 2019; T. Wood et al., 2020; Zhao et al., 2020). CO_2 adjustments typically manifest as a poleward expansion

10.1029/2023AV001144





Figure 5. Fast precipitation changes in response to a range of climate forcers in energy flux density units. For each forcer, fast precipitation changes are broken down into contributions from instantaneous changes, sensible heat and adjustments (left bar for each forcer). The adjustments are further decomposed into contributions from surface (T_s) , tropospheric (T_{Trop}) and stratospheric (T_{Strat}) temperature changes, water vapor and cloud changes (right bars). Based on Myhre et al. (2018) and Hodnebrog et al. (2020), who used multi-model simulations (Myhre et al., 2017) combined with radiative kernels. All numbers have been normalized by the multi-model mean ERF. BC has been further scaled by 0.2. The underlying experimental setups were $CO_2 \times 2$, $CH_4 \times 3$, insolation + 2% (SOLAR), BC × 10, $SO_4 \times 2$, $CFC-11 \times 8$, $CFC-12 \times 9$, $N_2O \times 3$.

of the extratropical circulation (Deser & Phillips, 2009; Grise & Polvani, 2014; T. Wood et al., 2020). This is thought to be related to enhanced meridional temperature gradients in the upper troposphere, although the detailed mechanisms remain unclear (Staten et al., 2014). Perturbations in stratospheric water vapor and ozone also have the potential to induce adjustments in large-scale circulation through altering meridional temperature gradients in the upper troposphere-lower stratosphere (e.g., Charlesworth et al., 2023; Maycock et al., 2013; Thompson & Solomon, 2002) though these composition changes may themselves occur as part of climate feedbacks (e.g., Nowack et al., 2015) or as an indirect response following the introduction of a forcing agent; for example, CO_2 driven stratospheric cooling alters ozone photochemistry, leading to tropospheric and stratospheric circulation adjustments (Chiodo & Polvani, 2016).

4. Adjustments Beyond the Earth Radiation Budget

In the main framework (Section 2), adjustments are considered as a modulating influence on the Earth radiation budget. However, adjustments can also manifest as perturbations to other climate parameters, notably precipitation, circulation patterns and lower atmosphere turbulence without immediate consequences for the Earth radiation budget. The influence of adjustments on precipitation can occur via any mechanism that affects the local tropospheric energy balance, or the atmospheric circulation. Recently, this has been investigated with climate models, for a range of climate individual drivers, by using a combination of radiative kernels (e.g., Kramer et al., 2019; Soden et al., 2008) and idealized experiments with Earth System Models (e.g., Hodnebrog et al., 2020; Myhre et al., 2018; Smith, Kramer, et al., 2018).

Precipitation change in response to a forcing can usefully be split into two components—in response to surface temperature change, called slow precipitation change, and in response to the initial forcing but without a change in surface temperature, called fast precipitation change (Bony et al., 2013). While the slow changes have been found to scale closely with ERF, which includes the effect of radiative adjustments (Andrews et al., 2010; Samset et al., 2016), the fast precipitation changes are adjustment processes that do not involve top-of-atmosphere radiation changes. Their underlying mechanisms are diverse and driver-specific. Figure 5 shows multi-model estimates of fast precipitation response that have been broken down into the influence of instantaneous changes, changes to sensible heat, and adjustments. This is possible because any precipitation change is constrained by the atmospheric energy balance, following the energetic relation for steady-state conditions (O'Gorman, 2012; Pendergrass & Hartmann, 2014):

$$L\Delta P = \Delta Q - \Delta S$$

(3)

Here, *L* is the latent heat of condensation, *P* is precipitation, *Q* is atmospheric radiative flux divergence (in general, cooling by emission of terrestrial radiation), and *S* is sensible heat. In Figure 5, the fast precipitation responses $(L\Delta P_{fast})$ are normalized by the ERF of the model experiment, in order to make them comparable. Note that this normalization is imperfect as ΔP_{fast} does not generally scale with ERF, but it still serves to provide consistent signs and to remove some of the differences in forcing strength between the model experiments used. Further, the adjustment contribution to ΔP_{fast} has been expanded into its contributions from surface, tropospheric and stratospheric temperature change, cloud changes, water vapor and clouds. These components are also scaled by the ERF, for comparability. Surface albedo changes can also contribute, but are negligible for all the experiments shown here and is therefore not shown.

When normalized by ERF, all components have negative total ΔP_{fast} . Broadly, adjustments are driven by a positive contribution from tropospheric temperature and cloud changes, counteracted by negative influences from surface temperature and water vapor. CO₂ change is the only climate forcer where the total adjustments are negative, which can be seen to be primarily due to the strong negative contribution from stratospheric cooling. The mechanism by which stratospheric cooling acts to reduce precipitation on fast time scales is via alteration of the atmospheric energy budget (Allen & Ingram, 2002; Bony et al., 2013).

The influence of adjustments on precipitation response is particularly strong for black carbon aerosols (Sand et al., 2020). Note that in Figure 5, the ΔP_{fast} from BC has been scaled by 0.2. Myhre et al. (2018) showed how BC is the driver with the highest model diversity, while Samset (2022) pointed out how aerosol absorption of solar radiation is very poorly constrained in the most recent set of coordinated Earth system model simulations—both in absolute magnitude, and in terms of changes over the historical era. The geographical pattern of the BC perturbation plays a large role (Dagan et al., 2019; A. I. L. Williams et al., 2023). Increases of BC concentrations in the Tropics lead to enhancement of precipitation and also precipitation extremes (Dagan & Eytan, 2024), while increases in the extratropics reduce precipitation. This makes further investigations into both the instantaneous and adjustment influence of BC on precipitation a key topic for constraining projections of precipitation under future emissions and climatic changes.

Changes in greenhouse gas and aerosol concentrations can also influence turbulence in the planetary boundary layer (PBL) independent of sea surface temperature changes. The influence of BC on PBL dynamics and turbulence is strongly dependent on the altitude of added BC (Slater et al., 2022; Wang et al., 2018). In many cases, however, the change in the vertical temperature profile following BC-induced heating causes an increase in stability, which leads to suppressed turbulence and a shallower PBL (Ding et al., 2016). This link between absorbing BC aerosols and suppressed turbulence has been identified from observations (Wilcox et al., 2016), case-studies using large-eddy simulation (LES) and regional modeling (Slater et al., 2022; Wang et al., 2018), and recently using global modeling (Stjern, Hodnebrog, et al., 2023). Furthermore, dimming by absorbing aerosol causes the downward solar radiation at the surface to decrease, resulting in changes of surface energy balances and turbulent exchange fluxes. As shown in Senf et al. (2021), in addition to stabilizing the PBL, this effect can weaken turbulence and convective development, negatively affecting the formation of shallow convective clouds (Feingold et al., 2005). Recent model results indicate that the fast response of scattering sulfate aerosols on lower tropospheric turbulence is weak, while the response of increased CO_2 concentrations is to enhance near-surface turbulence over land, opposite to that of BC (Stjern, Hodnebrog, et al., 2023). Changes in turbulence could influence exchange processes between the Earth's surface and the atmosphere, and human health through the dilution of air pollutants.

5. Observations of Adjustments

Observing adjustments is not straightforward. Most drivers of forcing change gradually rather than abruptly. Also, it is challenging to separate the adjustments in the observations from the meteorological variability, and it is also challenging to disentangle adjustments from IRF as well as from feedbacks. The time dependence of adjustments decouples them from the initial causal factor. At longer timescales, feedbacks and forced change of the climate system often dominate observed change (a notable exception is the stratospheric cooling in response to CO_2 , see Section 5.4). Nevertheless, there are cases where adjustments have been observed, typically in response to a specific perturbation, where the timescale of the change can be clearly determined and the response—at least, approximately—isolated from other factors.

5.1. Response to Volcanic Eruptions

Volcanic eruptions are an external perturbation to the climate system and are often well-defined. Large eruptions may affect the stratosphere and provide clues about stratospheric adjustments. On average, the stratosphere maintains an approximate balance between emission and absorption of radiation (temperature profile of radiative equilibrium; Section 1). If the emissivity or absorptivity of the stratosphere changes, then radiative balance is shifted to a new temperature. Changes in stratospheric composition can therefore give rise to stratospheric temperature adjustments. These adjustments can occur in response to direct injections of radiatively active gases or aerosols to the stratosphere, changes in the radiation incident on the stratosphere, or in response to changes in stratospheric cozone cools the stratosphere due to reduced up-welling radiation at the tropopause. If the stratospheric temperature adjustment is not uniform in space, then it can change stratospheric circulation.

Several examples of stratospheric temperature adjustments have been observed. The 1982 El Chichón and 1991 Pinatubo volcanic eruptions produced a layer of sulfate aerosol in the stratosphere that increased the absorption of longwave and solar near-infrared radiation. This caused significant stratospheric warming and tropospheric cooling that was detected in radiosonde (Labitzke & McCormick, 1992) and satellite observations (Robock, 2000). In addition to aerosol absorption (Stenchikov et al., 1998), heating, heterogeneous reactions on aerosol and changes in circulation combined to produce reductions in stratospheric ozone (Grant et al., 1992). The observed tropical stratospheric cooling of about 4 K following the 2022 Hunga Tonga-Hunga Ha'apai underwater eruption has been attributed to the large injection of water vapor into the stratosphere due to the eruption (Schoeberl et al., 2022).

Also the cloud microphysical adjustments to volcanic aerosol emissions into the troposphere may be observed and may serve to improve process understanding and as constraints for models (Malavelle et al., 2017). These analyses suggest a small change in cloud liquid water path (Haghighatnasab et al., 2022; Malavelle et al., 2017), but potentially a strong adjustment of cloud horizontal cover (Chen et al., 2022).

5.2. Response to Point Sources of Aerosols

Some of the clearest observations of adjustments come from the microphysical adjustments to localized aerosol emissions. Aerosol sources such as ships, effusive volcanoes and industry can produce observable changes in N_d , along with accompanying changes in macrophysical cloud properties (sometimes termed "opportunistic experiments," Christensen et al., 2022). Typically occurring as curvilinear polluted cloud features, the impact of the cloud microphysical adjustments are clear when compared to the relatively unpolluted clouds nearby. Increases in cloud height and decreases in precipitation have been observed in response to these aerosol perturbations (Christensen & Stephens, 2011), along with both increases and decreases in cloud water path (Gryspeerdt et al., 2019; Toll et al., 2017, 2019; Yuan et al., 2023). The spatially limited nature of these perturbations also highlights the time evolution of these adjustments (Goren & Rosenfeld, 2012; Gryspeerdt et al., 2021), evolving over hours to days. However, only a small fraction of the clouds polluted by shipping show visible ship tracks. It was recently shown that even in the case of ship track not distinguishable in the cloud microstructure, the clouds are affected by the ship emissions (Manshausen et al., 2022). This study found a positive adjustment of cloud liquid water path to increases in N_d , which may, however, be weak (Tippett et al., 2024).

Unique patterns of emissions also leave observable fingerprints of cloud microphysical adjustments at larger scales. The shipping corridor in the South-East Atlantic produces a clear change in N_d , but with less clear evidence of a change in cloud water. This implies a small role for cloud microphysical adjustments (Diamond et al., 2020). Enhanced lightning in the shipping corridors over the South China Sea and the northeastern Indian Ocean suggests a response of convective clouds to aerosol, with more vigorous convective activity at larger aerosol concentrations (Thornton et al., 2017).

5.3. Response to Wildfires

Emitting large amounts of absorbing aerosol, wildfires have the capability to alter atmospheric stability (Koch & Del Genio, 2010). Increases in smoke are correlated to reductions in cumulus cloud amount, through suppression of convective activity (Koren et al., 2004). In contrast, stability strengthening can increase cloud cover for stratocumulus clouds (Wilcox, 2010), expanding their cooling effect as a negative thermodynamic adjustment.

Recent extreme wildfires, such as the Australian pyroconvection events in 2019/2020, impressively demonstrated that local wildfire outbreaks can have global impacts (Senf et al., 2023; Yu et al., 2021). Much of the Australian smoke was emitted into the lower stratosphere, where it spread throughout the Southern Hemisphere after a short time (Ohneiser et al., 2020), leading to observable temperature increases in the stratosphere (Stocker et al., 2021) and dynamically driven adjustment processes (Kablick et al., 2020; Khaykin et al., 2020).

5.4. Trend Analysis of Recent Climate Change

Furthermore, anthropogenic emissions of greenhouse gases have increased the emissivity of the stratosphere and reduced the absorption of solar radiation by ozone, causing the stratosphere to cool. Significant long-term trends of stratospheric cooling have been detected in multi-decade satellite and radiosonde records (Maycock et al., 2018; Randel et al., 2009; Santer et al., 2023). The response of ozone concentrations to perturbations of the incoming solar radiation has been documented from satellite retrievals of the 11-year solar cycle (Dhomse et al., 2022; Hood et al., 2015; Maycock et al., 2016).

Short-lived climate forcers (SLCFs), such as aerosols and ozone, are spatio-temporally unevenly distributed. Thus, adjustments due to SLCFs are generally larger in regions and seasons with higher concentrations. It is expected that emissions of anthropogenic SLCFs will be further reduced as a result of air pollution control, but the progress of emission reductions much depends on the circumstances of each country. As countries are required to take action on climate change, a quantitative assessment of near-term climate change due to changes in emissions related to SLCFs by country or region is needed as a scientific basis. A quantification of adjustments is particularly important over such time scales. Since climate adjustments and feedbacks in a given country are not only due to emissions by this country and by a specific type of SLCFs, and since also impacts of specific emissions are not limited spatially, it is essential to consider multiple influencing factors to isolate adjustments attributable to specific emissions. There were attempts to analyze the microphysical adjustments to aerosol perturbations from the patterns of increasing versus decreasing aerosol emissions. The results are consistent with a small liquid water path adjustment but a positive relation between cloud fraction and drop number albeit with substantial uncertainty (e.g., Quaas et al., 2022).

6. Conclusions, Research Gaps, and Recommendations for Future Work

In the last decade, adjustments became an established part of the classical forcing—feedback framework to understand climate change from an energetic perspective. Substantial progress can be documented in terms of (a) analytical understanding, by classifying specific mechanisms by which adjustments act, (b) approaches to quantify adjustments across models and across the influences of specific state variable changes, and (c) the exploitation of observations to assess adjustment and constrain model estimates of these. An increasing number of studies aim for a more explicit analysis of adjustments that do not effect the Earth radiation budget immediately, but other components, like for example, precipitation fluxes.

Adjustments to changes in the Earth's orbit around the Sun, specifically the Milanković cycles that do not alter annual-, global-mean incoming solar radiation (changes to obliquity and precession of the orbit), are an evident manifestation of adjustments being relevant for large climate changes even in the absence of global instantaneous forcing.

Estimates of adjustments often come from idealized climate model simulations that employ large perturbations to, for example, greenhouse gas and aerosol concentrations, in order to improve signal-to-noise ratios. This leads to questions as to the linearity of adjustments and hence whether the size of adjustments found in such simulations are relevant for smaller, more realistic, forcings. Similar considerations apply to the additivity of adjustments when more than one climate forcing mechanism is present.

Climate model simulations can be used to understand the factors driving adjustments, to enable the development of more generic understanding of adjustment mechanisms (e.g., their dependence on the vertical profile of the instantaneous radiative heating perturbation), building on idealized studies including Hansen et al. (1997) and Salvi et al. (2021). An example of where this understanding would have great utility is in understanding the adjustments to halocarbon radiative forcing; there are a large number of halocarbon gases of potential climate importance, and it would be a challenge to accurately include these in climate model radiative transfer codes. To



date, adjustments have only been calculated for two CFCs (Hodnebrog et al., 2020) and the extent to which these adjustments are representative of a wider set of halocarbons is not known.

While this review has made a clear case that ERF is the preferred definition of forcing, the requirement to use climate models for the computation of adjustments does lead to some limitations. Climate model radiative transfer codes have limited spectral resolution, as well as requiring other approximations because of computational speed requirements; this necessitates continuing assessment of the quality of these codes for a wide range of forcings (including, separately, longwave and shortwave forcings) against benchmark radiation, via intercomparison projects such as the Radiative Forcing Model Intercomparison Project (Pincus et al., 2016). In addition, exploratory studies of new potential climate change mechanisms can be most easily done within an IRF or stratospheric adjusted radiative forcing framework using stand-alone radiative transfer codes, at the required spectral resolution, ahead of decisions on whether incorporation into climate model radiative codes is justified.

Adjustments to aerosol perturbations remain to be of particular importance. This is especially relevant for adjustments to aerosol-cloud interactions. Some aspects of these are especially poorly quantified such as the adjustments of deep convective clouds to aerosol-cloud interactions.

Adjustments are a sizable aspect in the forcing—feedback framework. Due to the typically rather short time scales on which they act, they lend themselves to constraints from observations more than feedbacks, despite the challenge in distinguishing IRF, adjustments and feedbacks in observations. Several examples for this are reported in this review. The analysis may investigate specific state variables (e.g., Figure 2) or specific adjustment mechanisms (Figure 1). In order to isolate, to better characterize, and to quantify specific mechanisms, one may analyze specific perturbations (as has been performed for aerosols), or one may use high-resolution, limited-area models thanks to the local nature in space and time of many adjustment processes (Nam et al., 2018). Model—data synergy arises when constraining models using observations, but also model studies may be used for a detection-attribution analysis by identifying fingerprints of adjustment mechanisms in models and assessing these in observations.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

No new data are produced for this review. The data for Figure 2 stems from previous publications by Hodnebrog et al. (2020), Marshall et al. (2020), and Skeie et al. (2020), and Smith, Kramer, et al. (2018). The data for Figure 3 were extracted from Doutriaux-Boucher et al. (2009), Zarakas et al. (2020), Mahowald (2011), Bellouin et al. (2020), Szopa et al. (2021), and Andrews et al. (2021). The data in Figure 4 are from Stjern, Forster, et al. (2023). The data for Figure 5, finally, are from Myhre et al. (2018) and Hodnebrog et al. (2020). For convenience, all data are tabulated and made available again in a concise way as Supporting Information S1.

References

- Abbott, T. H., & Cronin, T. W. (2021). Aerosol invigoration of atmospheric convection through increases in humidity. *Science*, 371(6524), 83–85. https://doi.org/10.1126/science.abc5181
- Ackerley, D., Chadwick, R., Dommenget, D., & Petrelli, P. (2018). An ensemble of AMIP simulations with prescribed land surface temperatures. *Geoscientific Model Development*, 11(9), 3865–3881. https://doi.org/10.5194/gmd-11-3865-2018
- Ackerman, A. S., Kirkpatrick, M. P., Stevens, D. E., & Toon, O. B. (2004). The impact of humidity above stratiform clouds on indirect aerosol climate forcing. *Nature*, 432(7020), 1014–1017. https://doi.org/10.1038/nature03174
 - Albrecht, B. A. (1989). Aerosols, cloud microphysics, and fractional cloudiness. *Science*, 245(4923), 1227–1230. https://doi.org/10.1126/science. 245.4923.1227
 - Allen, M., & Ingram, W. (2002). Constraints on future changes in climate and the hydrologic cycle. *Nature*, 419(6903), 224–232. https://doi.org/ 10.1038/nature01092
 - Andrews, T., & Forster, P. M. (2008). CO₂ forcing induces semi-direct effects with consequences for climate feedback interpretations. *Geophysical Research Letters*, 35(4), L04802. https://doi.org/10.1029/2007GL032273
 - Andrews, T., Forster, P. M., Boucher, O., Bellouin, N., & Jones, A. (2010). Precipitation, radiative forcing and global temperature change. *Geophysical Research Letters*, 37, 14. https://doi.org/10.1029/2010GL043991
 - Andrews, T., Gregory, J. M., Webb, M. J., & Taylor, K. E. (2012). Forcing, feedbacks and climate sensitivity in CMIP5 coupled atmosphere-ocean climate models. *Geophysical Research Letters*, 39(9), L09712. https://doi.org/10.1029/2012GL051607
 - Andrews, T., Smith, C. J., Myhre, G., Forster, P. M., Chadwick, R., & Ackerley, D. (2021). Effective radiative forcing in a GCM with fixed surface temperatures. *Journal of Geophysical Research: Atmospheres*, 126(4), e2020JD033880. https://doi.org/10.1029/2020JD033880
- Bala, G., Calderia, K., Mirin, A., Wickett, M., Delire, C., & Phillips, T. J. (2006). Biogeophysical effects of CO₂ fertilization on global climate. *Tellus B: Chemical and Physical Meteorology*, 58(5), 620–627. https://doi.org/10.1111/j.1600-0889.2006.00210.x

ically edback is for 'cific a sms, o' imited Mod ed for ing th (2018 Maho e 4 ar Kodne' Suppor humidit prescrit stratife (230. ht , 419(65)

RIGHTSLINK()

Acknowledgments

We gratefully acknowledge funding for this work by the EU Horizon 2020 project CONSTRAIN (GA no. 820829). E. G. was

supported by a Royal Society University Research Fellowship (URF/R1/191602). F.

M.O'C. acknowledges support from the

by DSIT and the Horizon 2020 project ESM2025 (GA no. 101003536), P.C.

Natural Environment Research Council

(NERC) Grants NE/T006250/1 and NE/ V012045/1. C.S. acknowledges support

from UK NERC Grant NE/T009381/1. K. S. acknowledges support from UK NERC

Grant NE/X004198/1. V.T. acknowledges support from the Estonian Research

acknowledges support from the Research Council of Norway Grant 275589. J.Q.

further acknowledges support by EU Horizon Europe project CleanCloud (GA

no. 101137639). J.M. was supported by

Office of Science, U.S. Department of

Regional and Global Model Analysis

program area. The Pacific Northwest

National Laboratory (PNNL) is operated for DOE by Battelle Memorial Institute

under contract DE-AC05-76RLO1830.

The authors would like to thank Brian

Open Access funding enabled and

organized by Projekt DEAL.

Soden, Bjorn Stevens and an anonymous

reviewer for their constructive comments.

Council Grant PRG1726. Ø.H.

Energy (DOE) Biological and Environmental Research as part of the

acknowledges support from the UK

Hadley Centre Climate Programme funded

- Bellouin, N., Quaas, J., Gryspeerdt, E., Kinne, S., Stier, P., Watson-Parris, D., et al. (2020). Bounding global aerosol radiative forcing of climate change. *Reviews of Geophysics*, 58(1), e2019RG000660. https://doi.org/10.1029/2019RG000660
- Bender, F., Frey, L., McCoy, D., Grosvenor, D. P., & Mohrmann, J. K. (2019). Assessment of aerosol-cloud-radiation correlations in satellite observations, climate models and reanalysis. *Climate Dynamics*, 52(7–8), 4371–4392. https://doi.org/10.1007/s00382-018-4384-z
- Bollasina, M., Ming, Y., & Ramaswamy, V. (2011). Aerosols and the weakening of the South Asian summer monsoon. Science, 334(334), 502– 505. https://doi.org/10.1126/science.1204994
- Bony, S., Bellon, G., Klocke, D., Sherwood, S., Fermepin, S., & Denvil, S. (2013). Robust direct effect of carbon dioxide on tropical circulation and regional precipitation. *Nature Geoscience*, 6, 447–451. https://doi.org/10.1038/ngeo1799
- Bony, S., Colman, R., Kattsov, V. M., Allan, R. P., Bretherton, C. S., Dufresne, J.-L., et al. (2006). How well do we understand and evaluate climate change feedback processes? *Journal of Climate*, 19(15), 3445–3482. https://doi.org/10.1175/JCLI3819.1
- Bony, S., & Stevens, B. (2020). Clouds and warming. In A. Siebesma, S. Bony, C. Jakob, & B. Stevens (Eds.), Clouds and climate: Climate science's greatest challenge (pp. 356–388). Cambridge University Press. https://doi.org/10.1017/9781107447738.012
- Bretherton, C. S. (2015). Insights into low-latitude cloud feedbacks from high-resolution models. *Philosophical Transactions of the Royal Society* A: Mathematical, Physical and Engineering Sciences, 373(2054), 20140415. https://doi.org/10.1098/rsta.2014.0415
- Bretherton, C. S., Blossey, P. N., & Uchida, J. (2007). Cloud droplet sedimentation, entrainment efficiency, and subtropical stratocumulus albedo. Geophysical Research Letters, 34(3), L03813. https://doi.org/10.1029/2006GL027648
- Burrows, S. M., McCluskey, C. S., Cornwell, G., Steinke, I., Zhang, K., Zhao, B., et al. (2022). Ice-nucleating particles that impact clouds and climate: Observational and modeling research needs. *Reviews of Geophysics*, 60(2), e2021RG000745. https://doi.org/10.1029/2021rg000745
- Charlesworth, E., Plöger, F., Birner, T., Baikhadzhaev, R., Abalos, M., Abraham, N. L., et al. (2023). Stratospheric water vapor affecting atmospheric circulation. *Nature Communications*, 14(1), 3925. https://doi.org/10.1038/s41467-023-39559-2
- Chemke, R., & Dagan, G. (2018). The effects of the spatial distribution of direct anthropogenic aerosols radiative forcing on atmospheric circulation. Journal of Climate, 31(17), 7129–7145. https://doi.org/10.1175/jcli-d-17-0694.1
- Chen, Y., Haywood, J., Wang, Y., Malavelle, F., Jordan, G., Partridge, D., et al. (2022). Machine learning reveals climate forcing from aerosols is dominated by increased cloud cover. *Nature Geoscience*, 15(8), 609–614. https://doi.org/10.1038/s41561-022-00991-6
- Chiodo, G., & Polvani, L. M. (2016). Reduction of climate sensitivity to solar forcing due to stratospheric ozone feedback. *Journal of Climate*, 29(12), 4651–4663. https://doi.org/10.1175/JCLI-D-15-0721.1
- Christensen, M. W., Gettelman, A., Cermak, J., Dagan, G., Diamond, M., Douglas, A., et al. (2022). Opportunistic experiments to constrain aerosol effective radiative forcing. Atmospheric Chemistry and Physics, 22(1), 641–674. https://doi.org/10.5194/acp-22-641-2022
- Christensen, M. W., & Stephens, G. L. (2011). Microphysical and macrophysical responses of marine stratocumulus polluted by underlying ships: Evidence of cloud deepening. *Journal of Geophysical Research*, *116*(D3), D03201. https://doi.org/10.1029/2010JD014638
- Chung, E.-S., & Soden, B. J. (2015). An assessment of methods for computing radiative forcing in climate models. *Environmental Research Letters*, 10(7), 074004. https://doi.org/10.1088/1748-9326/10/7/074004
- Colman, R., & McAvaney, B. (2011). On tropospheric adjustment to forcing and climate feedbacks. *Climate Dynamics*, 36(9–10), 1649–1658. https://doi.org/10.1007/s00382-011-1067-4
- Dagan, G. (2022a). Equilibrium climate sensitivity increases with aerosol concentration due to changes in precipitation efficiency. Atmospheric Chemistry and Physics, 22(24), 15767–15775. https://doi.org/10.5194/acp-22-15767-2022
- Dagan, G. (2022b). Sub-tropical aerosols enhance tropical cloudiness-a remote aerosol-cloud lifetime effect. Journal of Advances in Modeling Earth Systems, 14(12), e2022MS003368. https://doi.org/10.1029/2022ms003368
- Dagan, G., & Eytan, E. (2024). The potential of absorbing aerosols to enhance extreme precipitation. *Geophysical Research Letters*, 51(10), e2024GL108385. https://doi.org/10.1029/2024GL108385
- Dagan, G., Koren, I., Altaratz, O., & Heiblum, R. H. (2017). Time-dependent, non-monotonic response of warm convective cloud fields to changes in aerosol loading. *Atmospheric Chemistry and Physics*, 17(12), 7435–7444. https://doi.org/10.5194/acp-17-7435-2017
- Dagan, G., Stier, P., & Watson-Parris, D. (2019). Contrasting response of precipitation to aerosol perturbation in the tropics and extratropics explained by energy budget considerations. *Geophysical Research Letters*, 46(13), 7828–7837. https://doi.org/10.1029/2019g1083479
- Dagan, G., Yeheskel, N., & Williams, A. I. (2023). Radiative forcing from aerosol-cloud interactions enhanced by large-scale circulation adjustments. *Nature Geoscience*, 16(12), 1–7. https://doi.org/10.1038/s41561-023-01319-8
- DeMott, P. J., Prenni, A. J., Liu, X., Kreidenweis, S. M., Petters, M. D., Twohy, C. H., et al. (2010). Predicting global atmospheric ice nuclei distributions and their impacts on climate. *Proceedings of the National Academy of Sciences of the United States of America*, 107(25), 11217– 11222. https://doi.org/10.1073/pnas.0910818107
- Deser, C., & Phillips, A. S. (2009). Atmospheric circulation trends, 1950–2000: The relative roles of sea surface temperature forcing and direct atmospheric radiative forcing. *Journal of Climate*, 22(2), 396–413. https://doi.org/10.1175/2008JCL12453.1
- Dhomse, S. S., Chipperfield, M. P., Feng, W., Hossaini, R., Mann, G. W., Santee, M. L., & Weber, M. (2022). A single-peak-structured solar cycle signal in stratospheric ozone based on microwave limb sounder observations and model simulations. *Atmospheric Chemistry and Physics*, 22(2), 903–916. https://doi.org/10.5194/acp-22-903-2022
- Diamond, M. S., Director, H. M., Eastman, R., Possner, A., & Wood, R. (2020). Substantial cloud brightening from shipping in subtropical low clouds. AGU Advances, 1(1), e2019AV000111. https://doi.org/10.1029/2019av000111
- Dickinson, R. E., & Cicerone, R. J. (1986). Future global warming from atmospheric trace gases. Nature, 319(6049), 109–115. https://doi.org/10. 1038/319109a0
- Ding, A. J., Huang, X., Nie, W., Sun, J. N., Kerminen, V.-M., Petäjä, T., et al. (2016). Enhanced haze pollution by black carbon in megacities in China. Geophysical Research Letters, 43(6), 2873–2879. https://doi.org/10.1002/2016GL067745
- Dong, B., Gregory, J. M., & Sutton, R. T. (2009). Understanding land-sea warming contrast in response to increasing greenhouse gases. Part I: Transient adjustment. *Journal of Climate*, 22(11), 3079–3097. https://doi.org/10.1175/2009jcli2652.1
- Dong, B., Sutton, R. T., Highwood, E. J., & Wilcox, L. J. (2015). Preferred response of the East Asian summer monsoon to local and non-local anthropogenic sulphur dioxide emissions. *Climate Dynamics*, 46(5–6), 1733–1751. https://doi.org/10.1007/s00382-015-2671-5
- Doutriaux-Boucher, M., Webb, M. J., Gregory, J. M., & Boucher, O. (2009). Carbon dioxide induced stomatal closure increases radiative forcing via a rapid reduction in low cloud. *Geophysical Research Letters*, 36(2), L02703. https://doi.org/10.1029/2008GL036273
- Erb, M. P., Jackson, C. S., & Broccoli, A. J. (2015). Using single-forcing GCM simulations to reconstruct and interpret quaternary climate change. Journal of Climate, 28(24), 9746–9767. https://doi.org/10.1175/JCLI-D-15-0329.1
- Fan, J., Leung, L. R., Rosenfeld, D., Chen, Q., Li, Z., Zhang, J., & Yan, H. (2013). Microphysical effects determine macrophysical response for aerosol impacts on deep convective clouds. *Proceedings of the National Academy of Sciences of the United States of America*, 110(48), E4581– E4590. https://doi.org/10.1073/pnas.1316830110

- Feingold, G., Jiang, H., & Harrington, J. Y. (2005). On smoke suppression of clouds in Amazonia. Geophysical Research Letters, 32(2), L02804. https://doi.org/10.1029/2004GL021369
- Fels, S., Mahlman, J., Schwarzkopf, M., & Sinclair, R. (1980). Stratospheric sensitivity to perturbations in ozone and carbon dioxide: Radiative and dynamical response. *Journal of the Atmospheric Sciences*, 37(10), 2265–2297. https://doi.org/10.1175/1520-0469(1980)037(2265: SSTPIO)2.0.CO;2
- Forster, P. M., Freckleton, R., & Shine, K. (1997). On aspects of the concept of radiative forcing. *Climate Dynamics*, 13(7–8), 547–560. https:// doi.org/10.1007/s003820050182
- Forster, P. M., Richardson, T., Maycock, A. C., Smith, C. J., Samset, B. H., Myhre, G., et al. (2016). Recommendations for diagnosing effective radiative forcing from climate models for CMIP6. *Journal of Geophysical Research: Atmospheres*, 121(20), 12460–12475. https://doi.org/10. 1002/2016JD025320
- Forster, P. M., Storelvmo, T., Armour, K., Collins, W., Dufresne, J.-L., Frame, D., et al. (2021). The Earth's energy budget, climate feedbacks, and climate sensitivity. In V. Masson-Delmotte, et al. (Eds.), *Climate change 2021: The physical science basis. Contribution of working group I to the sixth assessment report of the intergovernmental panel on climate change* (pp. 923–1054). Cambridge University Press. https://doi.org/10. 1017/9781009157896.009
- Ganguly, D., Rasch, P. J., Wang, H., & Yoon, J.-H. (2012). Fast and slow responses of the South Asian monsoon system to anthropogenic aerosols. *Geophysical Research Letters*, 39(18), L18804. https://doi.org/10.1029/2012GL053043
- Geoffroy, O., Saint-Martin, D., Olivié, D. J. L., Voldoire, A., Bellon, G., & Tytéca, S. (2013). Transient climate response in a two-layer energybalance model. Part I: Analytical solution and parameter calibration using CMIP5 AOGCM experiments. *Journal of Climate*, 26(6), 1841– 1857. https://doi.org/10.1175/JCLI-D-12-00195.1
- Glassmeier, F., Hoffmann, F., Johnson, J. S., Yamaguchi, T., Carslaw, K. S., & Feingold, G. (2021). Aerosol-cloud-climate cooling overestimated by ship-track data. *Science*, 371(6528), 485–489. https://doi.org/10.1126/science.abd3980
- Goren, T., & Rosenfeld, D. (2012). Satellite observations of ship emission induced transitions from broken to closed cell marine stratocumulus over large areas. *Journal of Geophysical Research*, 117(D17), D17206. https://doi.org/10.1029/2012JD017981
- Grant, W. B., Fishman, J., Browell, E. V., Brackett, V. G., Nganga, D., Minga, A., et al. (1992). Observations of reduced ozone concentrations in the tropical stratosphere after the eruption of Mt. Pinatubo. *Geophysical Research Letters*, 19(11), 1109–1112. https://doi.org/10.1029/ 92GL01153
- Gray, L. J., Rumbold, S. T., & Shine, K. P. (2009). Stratospheric temperature and radiative forcing response to 11-year solar cycle changes in irradiance and ozone. Journal of the Atmospheric Sciences, 66(8), 2402–2417. https://doi.org/10.1175/2009JAS2866.1
- Gregory, J. M., Ingram, W. J., Palmer, M. A., Jones, G. S., Stott, P. A., Thorpe, R. B., et al. (2004). A new method for diagnosing radiative forcing and climate sensitivity. *Geophysical Research Letters*, 31(3), L03205. https://doi.org/10.1029/2003GL018747
- Gregory, J. M., & Webb, M. (2008). Tropospheric adjustment induces a cloud component in CO₂ forcing. *Journal of Climate*, 21(1), 58–71. https://doi.org/10.1175/2007JCL11834.1
- Grise, K. M., & Polvani, L. M. (2014). Is climate sensitivity related to dynamical sensitivity? A Southern Hemisphere perspective. *Geophysical Research Letters*, 41(2), 534–540. https://doi.org/10.1002/2013GL058466
- Grise, K. M., & Polvani, L. M. (2016). Is climate sensitivity related to dynamical sensitivity? Journal of Geophysical Research: Atmospheres, 121(10), 5159–5176. https://doi.org/10.1002/2015JD024687
- Gryspeerdt, E., Goren, T., & Smith, T. W. P. (2021). Observing the timescales of aerosol–cloud interactions in snapshot satellite images. Atmospheric Chemistry and Physics, 21(8), 6093–6109. https://doi.org/10.5194/acp-21-6093-2021
- Gryspeerdt, E., Goren, T., Sourdeval, O., Quaas, J., Mülmenstädt, J., Dipu, S., et al. (2019). Constraining the aerosol influence on cloud liquid water path. *Atmospheric Chemistry and Physics*, 19(8), 5331–5347. https://doi.org/10.5194/acp-19-5331-2019
- Gryspeerdt, E., Mülmenstädt, J., Gettelman, A., Malavelle, F. F., Morrison, H., Neubauer, D., et al. (2020). Surprising similarities in model and observational aerosol radiative forcing estimates. *Atmospheric Chemistry and Physics*, 20(1), 613–623. https://doi.org/10.5194/acp-20-613-2020
- Guo, L., Highwood, E., Shaffrey, L., & Turner, A. (2013). The effect of regional changes in anthropogenic aerosols on rainfall of the East Asian summer monsoon. Atmospheric Chemistry and Physics, 13(3), 1521–1534. https://doi.org/10.5194/acp-13-1521-2013
- Haghighatnasab, M., Kretzschmar, J., Block, K., & Quaas, J. (2022). Impact of Holuhraun volcano aerosols on clouds in cloud-system resolving simulations. Atmospheric Chemistry and Physics, 2(13), 8457–8472. https://doi.org/10.5194/acp-22-8457-2022
- Hansen, J. E., Sato, M., & Ruedy, R. (1997). Radiative forcing and climate response. *Journal of Geophysical Research*, *102*(D6), 6831–6864. https://doi.org/10.1029/96JD03436
- Hansen, J. E., Sato, M., Ruedy, R., Nazarenko, L., Lacis, A., Schmidt, G. A., et al. (2005). Efficacy of climate forcings. Journal of Geophysical Research, 110(D18), D18104. https://doi.org/10.1029/2005JD005776
- Hansen, J. E., Sato, M., Simons, L., Nazarenko, L. S., Sangha, I., Kharecha, P., et al. (2023). Global warming in the pipeline. Oxford Open Climate Change, 3(1), kgad008. https://doi.org/10.1093/oxfclm/kgad008
- Hodnebrog, O., Myhre, G., Kramer, R., Shine, K. P., Andrews, T., Faluvegi, G., et al. (2020). The effect of rapid adjustments to halocarbons and N₂O on radiative forcing. *Npj Climate and Atmospheric Science*, *3*(1), 43. https://doi.org/10.1038/s41612-020-00150-x
- Hood, L. L., Misios, S., Mitchell, D. M., Rozanov, E., Gray, L. J., Tourpali, K., et al. (2015). Solar signals in CMIP-5 simulations: The ozone response. *Quarterly Journal of the Royal Meteorological Society*, 141(692), 2670–2689. https://doi.org/10.1002/qj.2553
- IPCC. (1990). In J. Houghton, G. Jenkins, & J. Ephraums (Eds.), *Climate change: The IPCC scientific assessment*. Cambridge University Press. Retrieved from www.ipcc.ch/report/ar1/wg1
- Johnson, B. T., Haywood, J. M., & Hawcroft, M. K. (2019). Are changes in atmospheric circulation important for black carbon aerosol impacts on clouds, precipitation, and radiation? *Journal of Geophysical Research: Atmospheres*, 124(14), 7930–7950. https://doi.org/10.1029/ 2019JD030568
- Kablick, G., Allen, D. R., Fromm, M., & Nedoluha, G. (2020). Australian pyrocb smoke generates synoptic-scale stratospheric anticyclones. *Geophysical Research Letters*, 47(13), e2020GL088101. https://doi.org/10.1029/2020GL088101
- Kanji, Z. A., Welti, A., Corbin, J. C., & Mensah, A. A. (2020). Black carbon particles do not matter for immersion mode ice nucleation. *Geophysical Research Letters*, 47(11), e2019GL086764. https://doi.org/10.1029/2019gl086764
- Kärcher, B., & Lohmann, U. (2003). A parameterization of cirrus cloud formation: Heterogeneous freezing. *Journal of Geophysical Research*, 108(D14), 4402. https://doi.org/10.1029/2002jd003220
- Karset, I. H. H., Berntsen, T. K., Storelvmo, T., Alterskjær, K., Grini, A., Olivié', D., et al. (2018). Strong impacts on aerosol indirect effects from historical oxidant changes. Atmospheric Chemistry and Physics, 18(10), 7669–7690. https://doi.org/10.5194/acp-18-7669-2018
- Khaykin, S., Legras, B., Bucci, S., Sellitto, P., Isaksen, L., Tencé, F., et al. (2020). The 2019/20 Australian wildfires generated a persistent smokecharged vortex rising up to 35 km altitude. *Communications Earth & Environment*, 1, 22. https://doi.org/10.1038/s43247-020-00022-5



- Koch, D., & Del Genio, A. D. (2010). Black carbon semi-direct effects on cloud cover: Review and synthesis. Atmospheric Chemistry and Physics, 10(16), 7685–7696. https://doi.org/10.5194/acp-10-7685-2010
- Koren, I., Kaufman, Y. J., Remer, L. A., & Martins, J. V. (2004). Measurement of the effect of Amazon smoke on inhibition of cloud formation. *Science*, 303(5662), 1342–1345. https://doi.org/10.1126/science.1089424
- Koren, I., Remer, L., Altaratz, O., Martins, J., & Davidi, A. (2010). Aerosol-induced changes of convective cloud anvils produce strong climate warming. Atmospheric Chemistry and Physics, 10(10), 5001–5010. https://doi.org/10.5194/acp-10-5001-2010
- Kramer, R. J., Matus, A. V., Soden, B. J., & L'Ecuyer, T. S. (2019). Observation-based radiative kernels from CloudSat/CALIPSO. Journal of Geophysical Research: Atmospheres, 124(10), 5431–5444. https://doi.org/10.1029/2018JD029021
- Labitzke, K., & McCormick, M. P. (1992). Stratospheric temperature increases due to Pinatubo aerosols. *Geophysical Research Letters*, *19*(2), 207–210. https://doi.org/10.1029/91GL02940
- Leakey, A. D. B., Ainsworth, E. A., Bernacchi, C. J., Rogers, A., Long, S. P., & Ort, D. R. (2009). Elevated CO₂ effects on plant carbon, nitrogen, and water relations: Six important lessons from FACE. *Journal of Experimental Botany*, 60(10), 2859–2876. https://doi.org/10.1093/jxb/ erp096
- Lohmann, U. (2002). A glaciation indirect aerosol effect caused by soot aerosols. Geophysical Research Letters, 29(4), 11-1–11-4. https://doi.org/ 10.1029/2001gl014357
- Lohmann, U. (2017). Anthropogenic aerosol influences on mixed-phase clouds. Current Climate Change Reports, 3(1), 32–44. https://doi.org/10. 1007/s40641-017-0059-9
- Lohmann, U., Rotstayn, L., Storelvmo, T., Jones, A., Menon, S., Quaas, J., et al. (2010). Total aerosol effect: Forcing or radiative flux perturbation. Atmospheric Chemistry and Physics, 10(7), 3235–3246. https://doi.org/10.5194/acp-10-3235-2010
- Mahowald, N. (2011). Aerosol indirect effect on biogeochemical cycles and climate. Science, 334(6057), 794–796. https://doi.org/10.1126/ science.1207374
- Malavelle, F. F., Haywood, J. M., Jones, A., Gettelman, A., Clarisse, L., Bauduin, S., et al. (2017). Strong constraints on aerosol–cloud interactions from volcanic eruptions. *Nature*, 546(7659), 485–491. https://doi.org/10.1038/nature22974
- Manabe, S., & Wetherald, R. T. (1967). Thermal equilibrium of the atmosphere with a given distribution of relative humidity. *Journal of the Atmospheric Sciences*, 24(3), 241–259. https://doi.org/10.1175/1520-0469(1967)024(0241:TEOTAW)2.0.CO;2
- Manshausen, P., Watson-Parris, D., Christensen, M. W., Jalkanen, J.-P., & Stier, P. (2022). Invisible ship tracks show large cloud sensitivity to aerosol. *Nature*, 610(7930), 101–106. https://doi.org/10.1038/s41586-022-05122-0
- Marshall, L. R., Smith, C. J., Forster, P. M., Aubry, T. J., Andrews, T., & Schmidt, A. (2020). Large variations in volcanic aerosol forcing efficiency due to eruption source parameters and rapid adjustments. *Geophysical Research Letters*, 47(19), e2020GL090241. https://doi.org/10.1029/2020GL090241
- Maycock, A. C., Joshi, M. M., Shine, K. P., & Scaife, A. A. (2013). The circulation response to idealized changes in stratospheric water vapor. Journal of Climate, 26(2), 545–561. https://doi.org/10.1175/JCLI-D-12-00155.1
- Maycock, A. C., Matthes, K., Tegtmeier, S., Thiéblemont, R., & Hood, L. (2016). The representation of solar cycle signals in stratospheric ozone —Part 1: A comparison of recently updated satellite observations. Atmospheric Chemistry and Physics, 16(15), 10021–10043. https://doi.org/ 10.5194/acp-16-10021-2016
- Maycock, A. C., Randel, W. J., Steiner, A. K., Karpechko, A. Y., Christy, J., Saunders, R., et al. (2018). Revisiting the mystery of recent stratospheric temperature trends. *Geophysical Research Letters*, 45(18), 9919–9933. https://doi.org/10.1029/2018GL078035
- Mercado, L., Bellouin, N., Sitch, S., Boucher, O., Huntingford, C., Wild, M., & Cox, P. (2009). Impact of changes in diffuse radiation on the global land carbon sink. *Nature*, 458(7241), 1014–1017. https://doi.org/10.1038/nature07949
- Morgenstern, O., O'Connor, F. M., Johnson, B. T., Zeng, G., Mulcahy, J. P., Williams, J., et al. (2020). Reappraisal of the climate impacts of ozone-depleting substances. *Geophysical Research Letters*, 47(20), e2020GL088295. https://doi.org/10.1029/2020GL088295
- Murray, B. J., Carslaw, K. S., & Field, P. R. (2021). Opinion: Cloud-phase climate feedback and the importance of ice-nucleating particles. Atmospheric Chemistry and Physics, 21(2), 665–679. https://doi.org/10.5194/acp-21-665-2021
- Murray, B. J., O'sullivan, D., Atkinson, J., & Webb, M. (2012). Ice nucleation by particles immersed in supercooled cloud droplets. *Chemical Society Reviews*, 41(19), 6519–6554. https://doi.org/10.1039/c2cs35200a
- Murray-Watson, R. J., & Gryspeerdt, E. (2022). Stability-dependent increases in liquid water with droplet number in the Arctic. Atmospheric Chemistry and Physics, 22(9), 5743–5756. https://doi.org/10.5194/acp-22-5743-2022
- Myhre, G., Forster, P. M., Samset, B. H., Hodnebrog, O., Sillmann, J., Aalbergsjø, S. G., et al. (2017). PDRMIP: A precipitation driver and response model intercomparison project—Protocol and preliminary results. *Bulletin America Meteorology Social*, 98(6), 1185–1198. https:// doi.org/10.1175/BAMS-D-16-0019.1
- Myhre, G., Kramer, R. J., Smith, C. J., Hodnebrog, O., Forster, P., Soden, B. J., et al. (2018). Quantifying the importance of rapid adjustments for global precipitation changes. *Geophysical Research Letters*, 45(20), 11399–11405. https://doi.org/10.1029/2018GL079474
- Myhre, G., Shindell, D., Bréon, F.-M., Collins, W., Fuglestvedt, J., Huang, J., et al. (2013). Anthropogenic and natural radiative forcing. In T. Stocker, et al. (Eds.), *Climate change 2013: The physical science basis. Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change.* Cambridge University Press.
- Nam, C., Kühne, P., Salzmann, M., & Quaas, J. (2018). A prospectus for constraining rapid adjustments in general circulation models. *Journal of Advances in Modeling Earth Systems*, 10(8), 2080–2094. https://doi.org/10.1029/2017MS001153
- Nowack, P. J., Luke Abraham, N., Maycock, A. C., Braesicke, P., Gregory, J. M., Joshi, M. M., et al. (2015). A large ozone-circulation feedback and its implications for global warming assessments. *Nature Climate Change*, 5(1), 41–45. https://doi.org/10.1038/nclimate2451
- O'Connor, F. M., Abraham, N. L., Dalvi, M., Folberth, G. A., Griffiths, P. T., Hardacre, C., et al. (2021). Assessment of pre-industrial to presentday anthropogenic climate forcing in UKESM1. Atmospheric Chemistry and Physics, 21(2), 1211–1243. https://doi.org/10.5194/acp-21-1211-2021
- O'Connor, F. M., Johnson, B. T., Jamil, O., Andrews, T., Mulcahy, J. P., & Manners, J. (2022). Apportionment of the pre-industrial to present-day climate forcing by methane using UKESM1: The role of the cloud radiative effect. *Journal of Advances in Modeling Earth Systems*, 14(10), e2022MS002991. https://doi.org/10.1029/2022MS002991
- O'Gorman, P. (2012). Sensitivity of tropical precipitation extremes to climate change. *Nature Geoscience*, 5(10), 697–700. https://doi.org/10. 1038/ngeo1568
- Ohneiser, K., Ansmann, A., Baars, H., Seifert, P., Barja, B., Jimenez, C., et al. (2020). Smoke of extreme Australian bushfires observed in the stratosphere over Punta Arenas, Chile, in January 2020: Optical thickness, lidar ratios, and depolarization ratios at 355 and 532 nm. Atmospheric Chemistry and Physics, 20(13), 8003–8015. https://doi.org/10.5194/ACP-20-8003-2020
- Pendergrass, A. G., & Hartmann, D. L. (2014). The atmospheric energy constraint on global-mean precipitation change. *Journal of Climate*, 27(2), 757–768. https://doi.org/10.1175/JCLI-D-13-00163.1



- Persad, G., Samset, B. H., Wilcox, L. J., Allen, R. J., Bollasina, M. A., Booth, B. B. B., et al. (2023). Rapidly evolving aerosol emissions are a dangerous omission from near-term climate risk assessments. *Environmental Research: Climate*, 2(3), 032001. https://doi.org/10.1088/2752-5295/acd6af
- Persad, G. G., & Caldeira, K. (2018). Divergent global-scale temperature effects from identical aerosols emitted in different regions. Nature Communications, 9(1), 3289. https://doi.org/10.1038/s41467-018-05838-6
- Pincus, R., & Baker, M. (1994). Effects of precipitation on the albedo susceptibility of clouds in the marine boundary layer. *Nature*, 372(6503), 250–252. https://doi.org/10.1038/372250a0
- Pincus, R., Buehler, S. A., Brath, M., Crevoisier, C., Jamil, O., Franklin Evans, K., et al. (2020). Benchmark calculations of radiative forcing by greenhouse gases. *Journal of Geophysical Research: Atmospheres*, 125(23), e2020JD033483. https://doi.org/10.1029/2020jd033483
- Pincus, R., Forster, P. M., & Stevens, B. (2016). The radiative forcing model intercomparison project (RFMIP): Experimental protocol for CMIP6. Geoscientific Model Development, 9(9), 3447–3460. https://doi.org/10.5194/gmd-9-3447-2016
- Possner, A., Eastman, R., Bender, F., & Glassmeier, F. (2020). Deconvolution of boundary layer depth and aerosol constraints on cloud water path in subtropical stratocumulus decks. *Atmospheric Chemistry and Physics*, 20(6), 3609–3621. https://doi.org/10.5194/acp-20-3609-2020
- Quaas, J., Jia, H., Smith, C., Albright, A. L., Aas, W., Bellouin, N., et al. (2022). Robust evidence for reversal in the aerosol effective climate forcing trend. Atmospheric Chemistry and Physics, 22(18), 12221–12239. https://doi.org/10.5194/acp-22-12221-2022
- Ramanathan, V. (1975). Greenhouse effect due to chlorofluorocarbons—Climatic implications. *Science*, *190*(4209), 50–52. https://doi.org/10. 1126/science.190.4209.50
- Ramanathan, V., & Dickinson, R. E. (1979). The role of stratospheric ozone in the zonal and seasonal radiative energy balance of the Earthtroposphere system. *Journal of the Atmospheric Sciences*, 36(6), 1084–1104. https://doi.org/10.1175/1520-0469(1979)036(1084:TROSOI) 2.0.CO;2
- Ramanathan, V., Lian, M. S., & Cess, R. D. (1979). Increased atmospheric CO₂: Zonal and seasonal estimates of the effect on the radiation energy balance and surface temperature. *Journal of Geophysical Research*, 84(C8), 4949–4958. https://doi.org/10.1029/JC084iC08p04949
- Ramaswamy, V., Collins, W., Haywood, J., Lean, J., Mahowald, N., Myhre, G., et al. (2019). Radiative forcing of climate: The historical evolution of the radiative forcing concept, the forcing agents and their quantification, and applications. *Meteorological Monographs*, 59, 14.1–14.101. https://doi.org/10.1175/AMSMONOGRAPHS-D-19-0001.1
- Randel, W. J., Shine, K. P., Austin, J., Barnett, J., Claud, C., Gillett, N. P., et al. (2009). An update of observed stratospheric temperature trends. Journal of Geophysical Research, 114(D2), D02107. https://doi.org/10.1029/2008JD010421
- Reich, P. B., & Amundson, R. G. (1985). Ambient levels of ozone reduce net photosynthesis in tree and crop species. *Science*, 230(4725), 566–570. https://doi.org/10.1126/science.230.4725.566
- Richardson, T. B., Forster, P. M., Andrews, T., Boucher, O., Faluvegi, G., Fläschner, D., et al. (2018). Drivers of precipitation change: An energetic understanding. *Journal of Climate*, 31(23), 9641–9657. https://doi.org/10.1175/JCLI-D-17-0240.1
- Robock, A. (2000). Volcanic eruptions and climate. Reviews of Geophysics, 38(2), 191-219. https://doi.org/10.1029/1998RG000054
- Roeckner, E., Stier, P., Feichter, J., Kloster, S., Esch, M., & Fischer-Bruns, I. (2006). Impact of carbonaceous aerosol emissions on regional climate change. *Climate Dynamics*, 27(6), 553–571. https://doi.org/10.1007/s00382-006-0147-3
- Rosenfeld, D., Lohmann, U., Raga, G. B., O'Dowd, C. D., Kulmala, M., Fuzzi, S., et al. (2008). Flood or drought: How do aerosols affect precipitation? *Science*, 321(5894), 1309–1313. https://doi.org/10.1126/science.1160606
- Rotstayn, L. D., & Penner, J. E. (2001). Indirect aerosol forcing, quasi- forcing, and climate response. Journal of Climate, 14(13), 2960–2975. https://doi.org/10.1175/1520-0442(2001)014(2960:1AFQFA)2.0.CO;2
- Salvi, P., Ceppi, P., & Gregory, J. M. (2021). Interpreting the dependence of cloud-radiative adjustment on forcing agent. Geophysical Research Letters, 48(18), e2021GL093616. https://doi.org/10.1029/2021GL093616
- Samset, B. H. (2022). Aerosol absorption has an underappreciated role in historical precipitation change. *Communications Earth & Environment*, 3(1), 242. https://doi.org/10.1038/s43247-022-00576-6
- Samset, B. H., Myhre, G., Forster, P. M., Hodnebrog, O., Andrews, T., Faluvegi, G., et al. (2016). Fast and slow precipitation responses to individual climate forcers: A PDRMIP multimodel study. *Geophysical Research Letters*, 43(6), 2782–2791. https://doi.org/10.1002/ 2016GL068064
- Sand, M., Samset, B. H., Tsigaridis, K., Bauer, S. E., & Myhre, G. (2020). Black carbon and precipitation: An energetics perspective. Journal of Geophysical Research: Atmospheres, 125(13), e2019JD032239. https://doi.org/10.1029/2019JD032239
- Santer, B. D., Po-Chedley, S., Zhao, L., Zou, C.-Z., Fu, Q., Solomon, S., et al. (2023). Exceptional stratospheric contribution to human fingerprints on atmospheric temperature. Proceedings of the National Academy of Sciences of the United States of America, 120(20), e2300758120. https:// doi.org/10.1073/pnas.2300758120
- Schoeberl, M. R., Wang, Y., Ueyama, R., Taha, G., Jensen, E., & Yu, W. (2022). Analysis and impact of the Hunga Tonga-Hunga Ha'apai stratospheric water vapor plume. *Geophysical Research Letters*, 49(20), e2022GL100248. https://doi.org/10.1029/2022GL100248
- Seifert, A., Heus, T., Pincus, R., & Stevens, B. (2015). Large-eddy simulation of the transient and near-equilibrium behavior of precipitating shallow convection. *Journal of Advances in Modeling Earth Systems*, 7(4), 1918–1937. https://doi.org/10.1002/2015ms000489
- Senf, F., Heinold, B., Kubin, A., Müller, J., Schrödner, R., & Tegen, I. (2023). How the extreme 2019–2020 Australian wildfires affected global circulation and adjustments. *EGUsphere*, 2023, 1–28. https://doi.org/10.5194/egusphere-2023-113
- Senf, F., Quaas, J., & Tegen, I. (2021). Absorbing aerosol decreases cloud cover in cloud-resolving simulations over Germany. Quarterly Journal of the Royal Meteorological Society, 2021, 147(741), 4083–4100. https://doi.org/10.1002/qj.4169
- Shaw, T. A., & Voigt, A. (2015). Tug of war on summertime circulation between radiative forcing and sea surface warming. *Nature Geoscience*, 8(7), 560–566. https://doi.org/10.1038/ngeo2449
- Sherwood, S. C., Bony, S., Boucher, O., Bretherton, C. S., Forster, P. M., Gregory, J. M., & Stevens, B. (2015). Adjustments in the forcing-feedback framework for understanding climate change. *Bulletin America Meteorology Social*, 96(2), 217–228. https://doi.org/10.1175/ BAMS-D-13-00167.1
- Shindell, D. T., Faluvegi, G., Bell, N., & Schmidt, G. A. (2005). An emissions-based view of climate forcing by methane and tropospheric ozone. *Geophysical Research Letters*, 32(4), L04803. https://doi.org/10.1029/2004GL021900
- Shine, K. P., Cook, J., Highwood, E. J., & Joshi, M. M. (2003). An alternative to radiative forcing for estimating the relative importance of climate change mechanisms. *Geophysical Research Letters*, 30(20), 2047. https://doi.org/10.1029/2003GL018141
- Skeie, R. B., Myhre, G., Hodnebrog, Ø., Cameron-Smith, P. J., Deushi, M., Hegglin, M. I., et al. (2020). Historical total ozone radiative forcing derived from CMIP6 simulations. Npj Climate and Atmospheric Science, 3(1), 32. https://doi.org/10.1038/s41612-020-00131-0
- Slater, J., Coe, H., McFiggans, G., Tonttila, J., & Romakkaniemi, S. (2022). The effect of BC on aerosol-boundary layer feedback: Potential implications for urban pollution episodes. Atmospheric Chemistry and Physics, 22(4), 2937–2953. https://doi.org/10.5194/acp-22-2937-2022

- Small, J. D., Chuang, P. Y., Feingold, G., & Jiang, H. (2009). Can aerosol decrease cloud lifetime? *Geophysical Research Letters*, 36(16), L16806. https://doi.org/10.1029/2009g1038888
- Smith, C. J., Forster, P. M., Allen, M., Leach, N., Millar, R. J., Passerello, G. A., & Regayre, L. A. (2018). FAIR v1.3: A simple emissions-based impulse response and carbon cycle model. *Geoscientific Model Development*, 11(6), 2273–2297. https://doi.org/10.5194/gmd-11-2273-2018

Smith, C. J., Kramer, R. J., Myhre, G., Alterskjær, K., Collins, W., Sima, A., et al. (2020). Effective radiative forcing and adjustments in CMIP6 models. Atmospheric Chemistry and Physics, 20(16), 9591–9618. https://doi.org/10.5194/acp-20-9591-2020

- Smith, C. J., Kramer, R. J., Myhre, G., Forster, P. M., Soden, B. J., Andrews, T., et al. (2018). Understanding rapid adjustments to diverse forcing agents. *Geophysical Research Letters*, 45(21), 12023–12031. https://doi.org/10.1029/2018GL079826
- Soden, B. J., Held, I. M., Colman, R., Shell, K. M., Kiehl, J. T., & Shields, C. A. (2008). Quantifying climate feedbacks using radiative kernels. Journal of Climate, 21(14), 3504–3520. https://doi.org/10.1175/2007JCL12110.1
- Squires, P. (1952). The growth of cloud drops by condensation. I. general characteristics. Australian Journal of Chemistry, 5(1), 59. https://doi.org/10.1071/CH9520059
- Staten, P. W., Reichler, T., & Lu, J. (2014). The transient circulation response to radiative forcings and sea surface warming. *Journal of Climate*, 27(24), 9323–9336. https://doi.org/10.1175/JCLI-D-14-00035.1
- Stenchikov, G. L., Kirchner, I., Robock, A., Graf, H.-F., Antuña, J. C., Grainger, R. G., et al. (1998). Radiative forcing from the 1991 Mount Pinatubo volcanic eruption. *Journal of Geophysical Research*, 103(D12), 13837–13857. https://doi.org/10.1029/98JD00693
- Stevens, B., & Feingold, G. (2009). Untangling aerosol effects on clouds and precipitation in a buffered system. *Nature*, 461(7264), 607–613. https://doi.org/10.1038/nature08281
- Stevenson, D. S., Young, P. J., Naik, V., Lamarque, J.-F., Shindell, D. T., Voulgarakis, A., et al. (2013). Tropospheric ozone changes, radiative forcing and attribution to emissions in the atmospheric chemistry and climate model intercomparison project (ACCMIP). Atmospheric Chemistry and Physics, 13(6), 3063–3085. https://doi.org/10.5194/acp-13-3063-2013
- Stjern, C. W., Forster, P. M., Jia, H., Jouan, C., Kasoar, M. R., Myhre, G., et al. (2023). The time scales of climate responses to carbon dioxide and aerosols. *Journal of Climate*, 36(11), 3537–3551. https://doi.org/10.1175/JCLI-D-22-0513.1
- Stjern, C. W., Hodnebrog, O., Myhre, G., & Pisso, I. (2023). The turbulent future brings a breath of fresh air. *Nature Communications*, 14(3735), 3735. https://doi.org/10.1038/s41467-023-39298-4
- Stjern, C. W., Samset, B., Myhre, G., Forster, P. M., Hodnebrog, Ø., Andrews, T., et al. (2017). Rapid adjustments cause weak surface temperature response to increased black carbon concentrations. *Journal of Geophysical Research: Atmospheres*, 122(21), 11462–11481. https://doi.org/10. 1002/2017JD027326
- Stocker, M., Ladstädter, F., & Steiner, A. (2021). Observing the climate impact of large wildfires on stratospheric temperature. *Scientific Reports*, 11(1), 22994. https://doi.org/10.1038/S41598-021-02335-7
- Storelvmo, T., Kristjánsson, J. E., & Lohmann, U. (2008). Aerosol influence on mixed-phase clouds in CAM-Oslo. Journal of the Atmospheric Sciences, 65(10), 3214–3230. https://doi.org/10.1175/2008jas2430.1
- Szopa, S., Naik, V., Adhikary, B., Artaxo, P., Berntsen, T., Collins, W., et al. (2021). Short-lived climate forcers. In V. Masson-Delmotte, et al. (Eds.), Climate change 2021: The physical science basis. Contribution of working group I to the sixth assessment report of the intergovernmental panel on climate change (pp. 817–922). Cambridge University Press. https://doi.org/10.1017/9781009157896.008
- Tang, T., Shindell, D., Faluvegi, G., Myhre, G., Olivié, D., Voulgarakis, A., et al. (2019). Comparison of effective radiative forcing calculations using multiple methods, drivers, and models. *Journal of Geophysical Research: Atmospheres*, 124(8), 4382–4394. https://doi.org/10.1029/ 2018JD030188
- Thompson, D. W. J., & Solomon, S. (2002). Interpretation of recent Southern Hemisphere climate change. Science, 296(5569), 895–899. https:// doi.org/10.1126/science.1069270
- Thornhill, G. D., Collins, W. J., Kramer, R. J., Olivié, D., Skeie, R. B., O'Connor, F. M., et al. (2021). Effective radiative forcing from emissions of reactive gases and aerosols—A multi-model comparison. *Atmospheric Chemistry and Physics*, 21(2), 853–874. https://doi.org/10.5194/acp-21-853-2021
- Thornhill, G. D., Collins, W. J., Olivié, D., Skeie, R. B., Archibald, A., Bauer, S., et al. (2021). Climate-driven chemistry and aerosol feedbacks in CMIP6 Earth system models. Atmospheric Chemistry and Physics, 21(2), 1105–1126. https://doi.org/10.5194/acp-21-1105-2021
- Thornton, J. A., Virts, K. S., Holzworth, R. H., & Mitchell, T. P. (2017). Lightning enhancement over major oceanic shipping lanes. *Geophysical Research Letters*, 44(17), 9102–9111. https://doi.org/10.1002/2017GL074982
- Tippett, A., Gryspeerdt, E., Manshausen, P., Stier, P., & Smith, T. W. P. (2024). Weak liquid water path response in ship tracks. *EGUsphere*. Preprint. https://doi.org/10.5194/egusphere-2024-1479
- Toll, V., Christensen, M., Gassó, S., & Bellouin, N. (2017). Volcano and ship tracks indicate excessive aerosol-induced cloud water increases in a climate model. *Geophysical Research Letters*, 44(24), 12–492. https://doi.org/10.1002/2017g1075280
- Toll, V., Christensen, M., Quaas, J., & Bellouin, N. (2019). Weak average liquid-cloud-water response to anthropogenic aerosols. *Nature*, 572(7767), 51–55. https://doi.org/10.1038/s41586-019-1423-9
- Twomey, S. (1974). Pollution and the planetary albedo. Atmospheric Environment, 8(12), 1251–1256. https://doi.org/10.1016/0004-6981(74) 90004-3
- Vergara-Temprado, J., Holden, M. A., Orton, T. R., O'Sullivan, D., Umo, N. S., Browse, J., et al. (2018). Is black carbon an unimportant icenucleating particle in mixed-phase 10 clouds? *Journal of Geophysical Research: Atmospheres*, 123(8), 4273–4283. https://doi.org/10.1002/ 2017jd027831
- Voigt, A., Pincus, R., Stevens, B., Bony, S., Boucher, O., Bellouin, N., et al. (2017). Fast and slow shifts of the zonal-mean intertropical convergence zone in response to an idealized anthropogenic aerosol. *Journal of Advances in Modeling Earth Systems*, 9(2), 870–892. https:// doi.org/10.1002/2016ms000902
- Wang, Z., Huang, X., & Ding, A. (2018). Dome effect of black carbon and its key influencing factors: A one-dimensional modelling study. Atmospheric Chemistry and Physics, 18(4), 2821–2834. https://doi.org/10.5194/acp-18-2821-2018
- Wilcox, E. M. (2010). Stratocumulus cloud thickening beneath layers of absorbing smoke aerosol. *Atmospheric Chemistry and Physics*, 10(23), 11769–11777. https://doi.org/10.5194/acp-10-11769-2010
- Wilcox, E. M., Thomas, R. M., Praveen, P. S., Pistone, K., Bender, F. A.-M., & Ramanathan, V. (2016). Black carbon solar absorption suppresses turbulence in the atmospheric boundary layer. Proceedings of the National Academy of Sciences of the United States of America, 113(42), 11794–11799. https://doi.org/10.1073/pnas.1525746113
- Williams, A. I., Stier, P., Dagan, G., & Watson-Parris, D. (2022). Strong control of effective radiative forcing by the spatial pattern of absorbing aerosol. *Nature Climate Change*, 12(8), 735–742. https://doi.org/10.1038/s41558-022-01415-4
- Williams, A. I. L., Watson-Parris, D., Dagan, G., & Stier, P. (2023). Dependence of fast changes in global and local precipitation on the geographical location of absorbing aerosol. *Journal of Climate*, 36(18), 6163–6176. https://doi.org/10.1175/JCLI-D-23-0022.1



- Williams, E., Rosenfeld, D., Madden, N., Gerlach, J., Gears, N., Atkinson, L., et al. (2002). Contrasting convective regimes over the Amazon: Implications for cloud electrification. *Journal of Geophysical Research*, *107*(D20), 8082. https://doi.org/10.1029/2001JD000380
- Wood, R. (2007). Cancellation of aerosol indirect effects in marine stratocumulus through cloud thinning. *Journal of the Atmospheric Sciences*, 64(7), 2657–2669. https://doi.org/10.1175/JAS3942.1
- Wood, T., Maycock, A., Forster, P., Richardson, T. B., Andrews, T., Boucher, O., et al. (2020). The Southern Hemisphere midlatitude circulation response to rapid adjustments and sea surface temperature driven feedbacks. *Journal of Climate*, 33(22), 9673–9690. https://doi.org/10.1175/ jcli-d-19-1015.1
- Xie, X., Myhre, G., Liu, X., Li, X., Shi, Z., Wang, H., et al. (2020). Distinct responses of Asian summer monsoon to black carbon aerosols and greenhouse gases. Atmospheric Chemistry and Physics, 20(20), 11823–11839. https://doi.org/10.5194/acp-20-11823-2020
- Xie, X., Myhre, G., Shindell, D., Faluvegi, G., Takemura, T., Voulgarakis, A., et al. (2022). Anthropogenic sulfate aerosol pollution in South and East Asia induces increased summer precipitation over arid Central Asia. *Nature Communications Earth & Environment*, *3*(1), 1–8. https://doi.org/10.1038/s43247-022-00660-x
- Yu, P., Davis, S. M., Toon, O. B., Portmann, R. W., Bardeen, C. G., Barnes, J. E., et al. (2021). Persistent stratospheric warming due to 2019–2020 Australian wildfire smoke. *Geophysical Research Letters*, 48(7), e2021GL092609. https://doi.org/10.1029/2021GL092609
- Yuan, T., Song, H., Wood, R., Oreopoulos, L., Platnick, S., Wang, C., et al. (2023). Observational evidence of strong forcing from aerosol effect on low cloud coverage. *Science Advances*, 9(45), eadh7716. https://doi.org/10.1126/sciadv.adh7716
- Zarakas, C. M., Swann, A. L. S., Laguë, M. M., Armour, K. C., & Randerson, J. T. (2020). Plant physiology increases the magnitude and spread of the transient climate response to CO₂ in CMIP6 Earth system models. *Journal of Climate*, 33(19), 8561–8578. https://doi.org/10.1175/jcli-d-20-0078.1
- Zhang, J., Zhou, X., Goren, T., & Feingold, G. (2022). Albedo susceptibility of northeastern Pacific stratocumulus: The role of covarying meteorological conditions. Atmospheric Chemistry and Physics, 22(2), 861–880. https://doi.org/10.5194/acp-22-861-2022
- Zhao, X., Allen, R. J., Wood, T., & Maycock, A. C. (2020). Tropical belt width proportionately more sensitive to aerosols than greenhouse gases. Geophysical Research Letters, 47(7), e2019GL086425. https://doi.org/10.1029/2019GL086425